Contents lists available at ScienceDirect

Digital Engineering

journal homepage: www.sciencedirect.com/journal/digital-engineering

Human–machine interaction towards Industry 5.0: Human-centric smart manufacturing

Jialu Yang^a, Ying Liu^{a,*}, Phillip L. Morgan^b

^a Department of Mechanical Engineering, School of Engineering, Cardiff University, Cardiff, CF24 3AA, UK ^b Human Factors Excellence Research Group (HuFEx) and Centre for AI, Robotics and Human–Machine Systems (IROHMS), School of Psychology, Cardiff University, CF10 3AT, Cardiff, UK

ARTICLE INFO

Keywords: Human-machine interaction Industry 4.0 Industry 5.0 Sensor IoT IIoT Human-centric smart manufacturing Manufacturing informatics Data analytics

ABSTRACT

Since the concept of Industry 5.0 was proposed, the emphasis on human-machine interaction (HMI) in industrial scenarios has continued to increase. HMI is part of the factory's development towards Industry 5.0, mainly because HMI can help realise the human-centric vision. At the same time, to achieve the sustainable and resilient goals proposed by Industry 5.0, green, smart, and more advanced technologies are also considered important driving factors for factories to achieve Industry 5.0. Human-centric smart manufacturing (HCSM) factories that integrate HMI with advanced technologies are expected to become the paradigm of future manufacturing. Therefore, it is necessary to discuss technologies and research directions that may promote the implementation of HCSM in the future. In a smart factory, HMI signals will go through the process of being collected by sensors, processed, transmitted to the data analysis centre and output to complete the interaction. Based on this process, we divide HMI into four parts: sensor and hardware, data processing, transmission mechanism, and interaction and collaboration. Through a systematic literature review process, this article evaluates and summarises the current research and technologies in the HMI field and categorises them into four parts of the HMI process. Since the current usage scenarios of some technologies are relatively limited, the introduction focuses on the possible applications and problems they face. Finally, the opportunities and challenges of HMI for Industry 5.0 and HCSM are revealed and discussed.

1. Introduction

Artificial Intelligence (AI) has found its way into many aspects of daily life with applications like ChatGPT, Alexa and Face ID, but it has a much longer history in manufacturing. In the early 1980s, computer technology and automation began to be widely used in manufacturing. However, due to the limitations of computing power and hardware equipment, the ability to process data was relatively weak at that time [1,2]. Advances in these technologies lay the foundation for the realisation of smart factories. Smart factories have emerged with the rise of the Internet and the development of data storage technology. In the early 2000s, the industrial field began to adopt more advanced sensor technology and automation equipment, which meant that the datasets collected in the production process could be much larger and the data within them more accurate. Also, the development of cloud computing and big data technology provides a more optimal platform for processing and analysing these data [3,4]. Most previous smart factories focused on analysing process-related data. However, with the introduction of Industry 5.0, analysing human-related data to improve HMI may be a focus in future smart factories.

In 2011, the concept of Industry 4.0 was proposed at the Hannover Messe in Germany. It emphasises the interconnection of physical systems and digital systems. It realises a high degree of automation and intelligence in manufacturing through technologies such as digital twins, the Internet of Things, and data analysis [5–7]. This digital transformation improves production efficiency and flexibility and allows for personalised production, customised requirements and remote operation. This is inseparable from the development of smart factories.

Industry 5.0, however, takes the smart factory a step further and emphasises close cooperation between humans and machines, embracing human-centric, resilience and sustainability as core values [8]. It puts forward the vision of human-centric smart manufacturing (HCSM), advocates humanised smart manufacturing and focuses on human creativity, flexibility and professional knowledge [9]. This kind of manufacturing pursues human-machine collaborative work, combines the advantages of machines with human creativity, and promotes the sustainable development of industries.

To achieve the Industry 5.0 vision, it is essential to discuss humanmachine interaction (HMI). This is because, in Industry 5.0, the role of

* Corresponding author. E-mail address: LiuY81@Cardiff.ac.uk (Y. Liu).

https://doi.org/10.1016/j.dte.2024.100013

Received 18 April 2024; Received in revised form 4 July 2024; Accepted 26 July 2024 Available online 31 July 2024 2950-550X/© 2024 The Author(s). Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).



Review paper





many people within the factory/factories has changed and will continue to do so. This requires HMI to be better optimised to assist the operator efficiently. Machines will serve as extensions and assistants to help people make correct decisions and, at the same time, improve people's well-being through more humane assistance.

In Industry 4.0, the main research direction of HMI extended from the early automated psychological tests and knowledge acquisition to several major topics, including the distribution and allocation of tasks, trust and focus [10–15]. According to [16], the focal points and analytic approaches to HMI in Industry 4.0 focus on what can be divided into three taxonomies: human, machine, and interaction. Three-taxonomy focuses on the distinction of entities in the scene and the interactions between them. In the current paper, according to the process that HMI signals go through in manufacturing scenarios, HMI can be divided into four parts: (1) Sensors, where the signal is collected; (2) Data processing, where the signal and data are processed; (3) Transmission mechanism, how the data are transmitted, and (4) Interaction and collaboration. The purpose of adopting this classification is to better summarise the technologies involved in HMI rather than just focusing on research issues in the HMI field and the relationship between humans and machines. In Section 3, this paper will review the different types of technologies and research topics related to each part.

Although Industry 5.0 proposes a human-centric vision, the employees in the manufacturing scenario should be viewed from two perspectives [10]. First, people with professional capabilities will become the core of the HCSM factory. This is mainly because as the level of specialisation and intelligence of factories increases, the professional ability level of employees also needs to be correspondingly improved to operate these high-precision equipment. The HMI system should become their extensions and assistants, helping them better utilise their professional expertise and capabilities with a reduction (in areas) of physical aspects of work. On the other hand, the increase in integration and intelligence has also resulted in operators being unable to possess all manufacturing-related knowledge, meaning that they become "non-professional users" in some scenarios. Therefore, designing an empathetic, better interactive experience and personalised HMI system is the key to better serving these users. Key goals of HMI design in Industry 5.0 should be to reduce the burden on workers, improve their well-being and maximise their ability to realise their creativity.

Sustainability and resilience are the other two visions proposed by Industry 5.0, and they are also goals that the manufacturing industry has always wanted to achieve. In recent years, significant global factors such as pandemics and climate change have turned them into urgent issues and core values of Industry 5.0. The realisation of these visions cannot be separated from implementing new equipment and concepts. Developing optimal HMIs within this application domain and context is inseparable from these new technologies. VR and digital twins allow workers to monitor processes and perform operations anytime and anywhere, effectively improving the factory's resilience [17,18]. Machine learning (ML) and AI have greatly improved the efficiency of quality management, predictive maintenance, personnel planning and scheduling, thereby improving the sustainability of the factory. These all demonstrate the promotion of new technologies in manufacturing and HMI. The development of HMI in the future manufacturing industry will inevitably be inseparable from the application of these new technologies, and how these technologies optimally integrate with humans form the key focus points of the current article.

This paper aims to explore and study the opportunities and challenges that may be encountered in the future development of HMI in the manufacturing industry by summarising the technologies and topics in the HMI field. Section 1 introduces the background and motivation of the research. Section 2 presents the research methodology. Section 3 provides a review of research and technologies in the field of HMI. Section 4 discusses the opportunities and challenges that HMI will face in Industry 5.0. The final Section 5 summarises the topics reviewed in this paper and provides an outlook on possible future research directions.

Table 1

Research question.	
Research questions	Motivation
Q1: What are the main HMI-related technologies and research in the manufacturing field?	Analysing existing technologies and research can lead to a better understanding of their strengths and weaknesses.
Q2: What challenges will HMI face in HCSM and Industry 5.0, and how best to deal with them?	The human-centric concept is the core of Industry 5.0. Existing HMI equipment is still centred on efficiency, so it will inevitably face many new challenges.
Q3: What are the opportunities of HMI in HCSM and Industry 5.0?	As one of the important parts of realising the core value of Industry 5.0, the improvement and change that HMI can bring is one of the core issues we pay attention to.

Table 2

esearch database websites.	
Scopus	https://www.scopus.com
IEEE Xplore digital library	http://ieeexplore.ieee.org
ScienceDirect	http://www.sciencedirect.com
SpringerLink	https://link.springer.com
Google Scholar	https://scholar.google.com
ACM digital library	https://dl.acm.org

2. Methodology

To investigate the development of HMI in HCSM, this paper adopts a systematic literature review to identify and evaluate the relevant literature in this field. The overall approach is shown in Fig. 1. Our review process includes four main steps: defining research questions, identifying publication sources, identifying keywords, and paper selection. During the selection process, appropriate criteria are indispensable to ensure that papers provide a comprehensive overview of research in the field.

2.1. Research questions

This paper has three main research focuses: the current technologies and topics in HMI in manufacturing, the challenges that HMI will face in Industry 5.0, and the opportunities brought by implementing new technologies and concepts. To this end, we propose three research questions, as shown in Table 1.

2.2. The process of data collection

2.2.1. Search terms and resource identification

To better find research related to HMI in manufacturing, first, we need to determine the source of search resources and identify keywords.

The paper search is entirely done through the following scientific research databases: (see Table 2).

After identifying resource sources, we searched using keywords and included relevant literature references. There are two keyword searches. The first keyword search focused on an overview of HMI research in the manufacturing domain, thereby gaining a comprehensive understanding of the technologies involved. Search queries based on Boolean operators are as follows:

("Industry 5.0" OR "Industry 4.0" OR "Smart Manufacturing" OR "Human-centric Smart Manufacturing") AND ("Human-machine interaction" OR "human-robot interaction" OR "human-machine collaboration" OR "human-robot collaboration")

According to the results of this search, we added new keywords based on relevant review papers in this field containing terms that were not included within the original searches and conducted a second search. New keywords are HMI interface, Wearable Sensor, Computer



Fig. 1. Paper selection and evaluation process.

Vision, Optical Sensor, VR, XR, Hand-gesture Recognition, Acoustic Sensor, Speech Recognition, Anomalous Sound Detection, Bionic Sensor, Brain-computer Interface, Tactile Sensor, Motion Detection, Internet of Things, Internet of Everything, Big Data, Cloud Computing, Edge Computing, Cloud Computing Security, Machine Learning, Deep Learning, Wireless Sensor Network, Industrial Ethernet, 5G, 6G, data transmission, Function allocation, workload allocation.

2.2.2. Criteria for paper selection

The inclusion and exclusion criteria proposed within the systematic review were set based on the proposed aims and research questions. Exclusion criteria:

- 1. Papers that topics are not relevant to the manufacturing field. If the technologies and research in a paper are completed in a manufacturing scenario and can be applied to manufacturing, this paper will be marked as 'relevant'. If a paper is not about HMI in manufacturing but its main content can be applied to manufacturing, it will be marked as 'less relevant'. If a paper is not about manufacturing solutions and its core idea and model cannot be applied to manufacturing, it will be marked as 'not relevant'.
- 2. The paper must be available in English.
- 3. For technology papers, the publication time should be after 2019. This is mainly because the concept of Industry 5.0 was proposed in 2020.

2.3. Paper selection

The first keyword search aims to comprehensively understand the relevant technologies in this field. A total of 47 relevant results were obtained. Based on papers in this research, we conducted a second search with new and original keyword combinations. We eliminated the duplicated results of the two searches and finally identified 234 relevant and less relevant papers. Then, We compared the relevance between papers and our research question and excluded papers that discussed duplicate issues. Finally, 152 papers were included in the review after all criteria were applied.

3. Human-machine interaction in smart manufacturing

3.1. Overall framework of HMI

As mentioned in previous sections, according to [16], HMI in Industry 4.0 can be divided into three categories: Human, Machine and Interaction. Due to the different attributes, Humans and Machines are separated, and Interaction is classified separately as it is one of the most important parts of this process. Fig. 2 presents this taxonomy and some of the subcategories within each category. This classification method is simple and intuitive. However, it can be found that each major category's subcategories are very different. For example, machines include psychological topics such as roles and functions and hardware and technology. Their commonalities are relatively small and cannot be compared when discussed. Each subcategory in this classification method needs to be more cohesive, and this classification method emphasises the differences rather than the connections. With more interdisciplinary research expected in Industry 5.0, this three-category taxonomy needs to be revised, as it fails to represent the HMI relationship in a highly integrated, closely connected smart factory.

To this end, we propose a framework to classify HMI according to the data transmission process in the factory, as shown in Fig. 2. Sensors are the first step in the HMI process in a smart factory, and signals are fed into the system through these devices. These signals and data are then sent to the data centre for processing and analysis, and humans in the data centre will make decisions with the help of machines. The input of data and the output of instructions are inseparable from the transmission mechanism, and this part also requires the cooperation of various hardware. Finally, after receiving the instruction, the device completes the interaction with the operator [19]. This paper briefly introduces the enabling technologies and related research designed in each part.

3.2. Sensor

Sensors in manufacturing scenarios can be divided into five categories according to different principles: optical-based, acoustic-based, bionic-based, tactile-based and motion-based sensors [19]. In smart factories, most sensors mainly collect the environment and processrelated data. This is mainly based on two considerations. First, Industry 4.0 is efficiency-driven [8]. Second, collecting data related to people can easily violate workers' privacy.

To better understand the scenarios used by these sensors, the HMI interface must first be introduced. HMI interfaces in the industry can be divided into two categories [20]. The first category is the physical part of a machine that a user can see and touch, such as a computer interface. The other category is invisible natural user interfaces (NUIs), where users use intuitive everyday behaviours to perform interactions, such as gesture recognition [21]. Their difference is that physical interaction requires prior operator training, while NUI focuses on understanding user behaviours.

According to different interaction methods, HMI in smart factories can also be divided into five categories: gaze, voice, gesture, tactile,



Fig. 2. HMI's three-category and four-category taxonomies.

and haptic interactions [2]. Gaze interaction mainly interacts with devices that detect human eyes [22]. Voice completes the control of the device by inputting voice information [23]. Gesture recognition can not only collect hand movement data through cameras but also collect data through wearable devices to complete the interaction [19,20]. Tactile is closely related to haptic interaction. Tactile refers to operators performing tasks through tablets and physical interfaces. In contrast, haptic recognition focuses more on nonverbal communication (sound, vibration, touch, and temperature) provided by wearable devices [2].

Next, we comprehensively introduce and discuss each sensor and its interaction interface and method.

3.2.1. Optical-based sensors

Optical sensors do not require physical contact and, as such, are widely used in public interfaces and NUIs [19]. LEDs, lasers and camera–computer vision are three of the most common types of opticalbased sensors. LEDs and lasers are often used for position detection and recognition in traditional manufacturing, which makes them also applicable to gesture detection. They are also used in multi-touch sensors and displays [19,24]. Their structures are often relatively simple, with low cost and convenient deployment advantages. Fields such as motion detection, hand gesture recognition (HGR) and eye tracking are inseparable from the application of camera–computer recognition. It is also one of the technologies that enable extended reality (XR) (virtual reality (VR), augmented reality (AR), and mixed reality (MR)) [20].

Since the 1980s, VR glasses have been used to train pilots [25]. With the continuous development of technology in this field, VR headmounted displays (HMDs) have been widely used in various fields [18, 26]. As the demand for customised products continues to grow, the need for production processes and varieties also grows, further increasing the need for operators. The study found that in High-Mix Low-Volume (HMLV) assembly systems and Mixed Model Assembly Lines (MMALs), complex and variable operating systems and environments lead to a decline in worker efficiency and reduce worker satisfaction [27]. An effective way to solve this problem is XR. XR reduces the cost of understanding for workers and can better deliver effective information, making the workplace a 'small-batch, knowledgedriven place' [17,28]. However, some research results show that XR does not always assist workers in reducing workload in industrial scenarios [29]. The reasons for this problem can be summarised into four points: 1. XR cannot assist in low-complexity work [30]. 2. Current HMDs are still bulky and cannot be used long. Therefore, better results can often be obtained in spatial and projection-based XR applications [31]. 3. Interface also greatly impacts workers' understanding of the system, and NUI is considered to have lower learning costs in many

cases [20,21,32]. 4. The information presentation mode will also affect the effect of XR, such as annotation, information availability and visual interference [31].

Vision-based HGR uses cameras to capture scene information and machine learning to analyse gesture features. Such systems generally consist of three components: detection, tracking and recognition. In the detection phase, the camera captures images of hand gestures and segments them from the background [33,34]. In the tracking process, the computer will continuously analyse the characteristics of the hand to realise real-time dynamic recognition [35]. Finally, through machine learning algorithms, the computer recognises and classifies gestures [36]. Generally, ordinary RGB (red, green and blue) cameras on the market can complete the HGR task. However, the improvement of the quality of the camera can not only improve the recognition accuracy but also require the device to have better computing power [34]. Therefore, selecting the most suitable sensor according to the task is also a major issue that needs to be considered when designing the HGR system. Kinect and Leap Motion are widely used products with high reliability and complete functions in vision-based HGR design. Kinect achieved an average recognition ability of 95% and 98.9% in recognising numbers and letters written in the air [34,37]. Leap Motion has been experimentally verified to improve the accuracy of intelligent systems and robot interactions [38].

Over the years, many researchers have continued to improve visual sensors to make them perform better, be more portable, and be suitable for more scenarios. Part of the research mainly focuses on reducing the burden on people who have existing equipment through new materials and designs. Nanoparticles, nanowires, carbon nanotubes (CNTs), and graphene have been used to build new types of XR devices [18]. They are characterised by strong stretchability, flexibility, and low load, so they can effectively solve the problems of bulky and poor flexibility of many HMDs [39]. In addition, smart contact lenses are also a topic that shows great potential [19]. In terms of medical treatment, it has been used in vision aids [40], corneal temperature detection [41], intraocular pressure detection [42], blood sugar level detection [34] and other fields, and has shown good results. It is used in medical, visual aids, near-eye displays and other fields and has great eye-tracking potential. In addition, it may be an effective solution to the damage to vision caused by VR HMDs.

3.2.2. Acoustic-based sensors

In Industry 4.0, acoustic sensors are widely used in machine status detection, fault detection, and voice recognition [19,43]. Compared with contact sensors such as probes and temperature sensors, acoustic sensors have the advantage of low cost and do not require physical

contact with the sensor. However, due to the large amount of noise in most industrial environments, the use or quality of the signals and data produced by this type of sensor is limited. Abnormal sound detection is the basis of machine status and fault detection [44]. However, since the abnormal sound data is relatively small compared with the overall data (data imbalance problem), and these data are various, it is not easy to put these data into the machine learning algorithm for training [45]. For this reason, some researchers have adopted a multi-microphone serial combination system to suppress noise and reverberation [46]. In addition to reducing the impact of noise by adding sensors, algorithms such as Autoencoder-Based [43,45], Gaussian Mixture Model-Based [47] and Outlier Exposure-Based [48] anomaly detection are also used to eliminate noise and improve data imbalance problems.

Speech recognition technology has greatly progressed with AI development [49]. A speech recognition system consists of acoustic hardware sensors and speech recognition software [44]. Condenser microphones are one of the representatives of acoustic sensors. They use capacitance differences, such as voltage changes, to detect sound signals [50]. However, this sensor has disadvantages such as instability, short recognition distance, high power consumption, and low sensitivity. A piezoresistive microphone detects sound through a change in resistance. Its advantage over capacitive sensors is that they can still detect sound at high temperatures because the resistance is temperature-dependent [51]. The advantage of a triboelectric sensor is that it does not require an external power supply [44,52]. However, due to electrostatic phenomena, such sensors are easily affected by humidity and temperature. Flexible piezoelectric sensors that mimic the structure of the human cochlea are considered suitable candidates for speech recognition due to their high sensitivity and recognition rate [51]. It has a self-powered structure and can perform well in complex environments due to bionic concepts and flexible film materials [53]. The sensor has proven 97.5% accurate in recognising speech [44].

3.2.3. Bionic-based sensors

Bionic sensors detect biological signals through computer technology and biotechnology, and they generally have the characteristics of ultra-sensitivity, self-adaptation, and low power consumption [19]. Bionic materials, bionic structures, and functional bionic are three bionic strategies for sensors [54]. Bionic materials are components in sensors that use materials that mimic the structure of natural materials [55]. This gives these sensors unique properties. Bionic structures mean sensors realise certain functions and characteristics by imitating and learning biological and natural structures. The waterproof super-hydrophobic film is obtained by imitating super-wet biomimetic materials [56]. The difference with functional bionics is that it mainly improves sensor detection capabilities by imitating biological senses rather than focusing on adding new functions [19].

Generally, collecting biological signals by bionic sensors is completed through electrodes. The electrodes can collect different signals by changing the monitoring frequency and level amplification [19]. According to different sources, these biological signals can come from the brain, muscles and movements [57]. The electroencephalographybased brain-computer interface and the electromyography-based myoelectric interaction are two of the more common ones. Electrooculography and electrocardiograms have also received continuous attention in recent years [19,54]. Brain-computer interface is one of the most concerned studies in this field, but due to ethical and moral issues, research in this field is somewhat controversial. In the industrial field, braincomputer interfaces can already be used for robotic arm control [58]. This technology will play an important role in HCSM because it can meet the needs of Industry 5.0 for sensors to serve humans. Like the brain-computer interface, the electromyographic interaction completes the control process by detecting the muscles' electrical signals. A major application direction of this technology is exoskeleton and auxiliary limbs [59].

3.2.4. Tactile-based sensors

Tactile sensors include all sensing devices that require physical contact, such as operable buttons, tablet computers (tactile), and non-verbal communication (haptic) provided by wearable devices. According to different principles, these sensors can be divided into capacitive, piezo-resistive, piezoelectric and optical tactile sensors [60]. The difference in principle determines the difference in the materials they use. Among them, the material of the optical tactile sensor must have optical transparency and elasticity.

The main purpose of tactile sensing is to obtain information about the environment and objects to manipulate them [61]. Therefore, much of the early research on this type of sensor was focused on the medical field. For example, compared with traditional surgery, tactile sensorbased surgery involves smaller incisions, less blood loss, and is safer overall [62]. However, the performance of these devices depends on the accuracy of the tactile information collected [61]. In the manufacturing industry, the force sensor represented by the Telerobot hand pressure sensor is one of the main application scenarios of the tactile sensor [19]. In recent years, as robots have become more and more intelligent and anthropomorphic, basic sensory system simulation is also a research direction that has attracted wide attention. By implementing simulation of basic perception systems, robots can become more like humans and have stronger empathy, which is more conducive to human-machine collaboration. In the robot grasping problem, the combination of tactile and visual information is a problem many researchers pay attention to [63].

Tactile sensors are also the basis of wearable devices. In VR devices, these sensors can collect haptic information such as vibration, touch, and temperature to provide users with more realistic and sensitive feedback [2,33,34]. In addition, these tactile sensors have become smaller and more sensitive through the combination with bionic technology, making them have a wider range of application scenarios [54].

3.2.5. Motion-based sensors

Motion sensors in HMIs, such as accelerometer pointers and gyroscopes, are often used in wearable devices to detect the motion of objects [19]. The function and structure of these sensors are relatively simple. Some studies have used them as an alternative to optical sensors because of their lower space requirements [64]. Most studies use a combination of gyroscopes, magnetic needles, and accelerometers to determine the 3D position of the object of interest — including humans [65,66]. A recent review paper [65] concludes that healthcare is the most common application area for inertial motion sensors. These sensors improve human physiological conditions by collecting data on human movement [64]. Although some studies used two types of motion sensors to analyse 2D planar data, most studies used 3D data. In the industrial field, motion sensors have been used in equipment and production line detection [67].

As one of the most important parts of HMI, sensors are responsible for collecting information and signals and completing interactions. The design of these sensors not only affects the efficiency and ability to collect signals but also affects the operator's experience. A good sensing system should be able to collect valid data, be easy to use, and not place additional burdens on workers.

3.3. Data processing

Since the sensor(s) collects only raw signals, these signals must be processed to obtain the desired results. The original dataset collected by the sensor cannot be processed directly because the quality and quantity of the data may not meet the requirements of the data analysis algorithm [5]. In this case, the original data set must be pre-processed. Reduced noise data, feature extraction, and data fusion are common methods for processing and integrating data in some industries [68]. Taking appropriate processing methods can improve the data's validity and the model's accuracy.

Three types of problems may occur in the data collected in the industry: too much data, too little data, and fractured data [68]. Too much data is very common in the manufacturing industry. This is mainly because the current manufacturing environment is often equipped with many sensors, which will collect a large amount of data. However, not all of this data is valid. Noise, irrelevant, and categorical/numerical mixed data exist [69]. When splitting the dataset, too little data may be caused by missing attributes or uneven distribution of features [68,69]. In this case, the reliability of the model is very low. When data comes from multiple sources and platforms, it may lead to fractured data due to incompatibility. In addition, the different weight levels of data in the model and repository may also cause this problem [68,70].

After collecting the data, we first need to determine the appropriate preprocessing method based on the problems in the data. Reducing noise data is a method that most original datasets need to adopt. Binning, regression and clustering are three common methods to reduce noise data [69]. Binning can replace data with smoother interval discrete values to reduce observation errors [68,71]. Regression uses machine learning to predict variance and bias variables to improve accuracy. Robust regression, Gaussian process regression, support vector regression, polynomial regression, linear regression and decision trees are some common regression analyses. Clustering analyses the intrinsic connections between data and groups them to exclude noise and irrelevant data. Some common clustering methods are k-means, spatial clustering based on noise applications, and mean shift [68]. In addition, feature extraction can also be used to filter data. The features of data collected in the manufacturing industry include process features, process features, personnel features, etc. [70]. Statistical and regression analysis, dimensionality reduction analysis, text mining and image processing are common feature extraction methods [70,72]. Data fusion can solve the fractured data problem and increase the size of the data set, so it is also an important processing method in the manufacturing industry. Bayes 'Rule, Probabilistic Grids, The Kalman Filter, Sequential Monte Carlo Methods and Alternatives to Probability are common multi-sensor data fusion methods [73]. Bayes 'Rule infers changes in environment and state values by building a probabilistic model. Probabilistic Grid is a method that implements Bayesian data fusion through mapping. The Kalman Filter is a Bayesian filter that uses a recursive function to calculate changes in continuous values. Sequential Monte Carlo Methods can convert probability distributions into weighted samples in space for calculation. Interval calculus, fuzzy logic and evidence reasoning are common Alternatives to Probability methods. They can solve the complexity, inconsistency, uncertainty and precision problems in probabilistic modelling [73].

After the data pre-processing, the smart factory will analyse the data set and build a model to achieve predictive maintenance and condition monitoring goals. In this analysis process, most industrial scenarios have relatively high requirements for real-time instructions, so there are requirements for the factory's data transmission and processing capabilities. Here, smart factories need suitable data processing solutions. This section will start with the Internet of Things (IoT) and introduce cloud computing (CC), edge computing (EC), and machine learning and their place in smart factories.

3.3.1. Internet of things and internet of everything

The concept of IoT can be traced back to the late 1990s. Its main goal is to control processes by connecting devices, sensors and software [74]. By connecting various devices to the network, not only can we realise functions such as predictive maintenance, production monitoring, supply chain optimisation, energy saving, inventory management, and quality tracking, but also, the data collected in the factory can be effectively integrated to maximise the use of the potential of this data [7]. IoT has three enabling technologies [75]:

• Radio Frequency Identification (RFID) and Near Field Communication (NFC) technology

- · Wireless Sensor Networks (WSN) technology
- Data Storage and Analytics technology

RFID technology guarantees real-time traceability, controllability and visibility of personnel, materials and equipment [1]. NFC is built based on RFID, which ensures the transmission and reception of short-distance data [75]. WSN will be introduced more specifically in the next section. Data Storage and Analytics technology development is inseparable from CC, machine learning and the establishment of big data platforms [19]. These technologies are necessary to extract useful information from collected big data.

There are six main functional elements of IoT: semantics, identification, sensing, services, computational and communication [76]. Among them, identification realises the naming and matching of elements in the network. Subsequently, the sensing element collects data and sends it to the computing centre. The key to realising digital services for different objects is communication. The computational and semantics functions provided by both hardware and software ensure that the system can send signals and instructions to various components in realtime to achieve the ultimate goal of IoT: to provide services for anyone at any time.

After years of development, IoT has been widely used. These applications mainly focus on five areas: Smart City, Internet of Medical Things (IoMT), Smart Grid, Internet of Vehicles (IoV), and Industrial Internet of Things (IIoT).

IoT-based smart cities collect data through different sensors and identification technologies and implement energy management, road maintenance, water supply systems online, crime prevention, community planning and other functions based on these data [77,78]. IoT-based smart cities can provide more timely, intelligent and high-quality services than the current urban management system. However, since these data often involve residents' privacy, data security is a major issue in smart city deployment.

The rapid development of IoMT and travel restrictions brought about by COVID-19 are inseparable. IoMT enables medical professionals and hospitals, more generally, to understand patients' data in real-time through remote devices and wireless medical devices to provide timely medical treatment [79]. The often very small wearable device(s) can transmit the patient's heartbeat, blood oxygen, blood pressure and other health data to the medical centre, which enables people to receive continuous health monitoring support. However, personal health data breaches will hit 45 million worldwide in 2021, tripling in just three years, according to a report by cybersecurity firm Critical Insights [76]. Since patient information and their identities are closely linked in the IoMT, these leaks can greatly impact patients' treatment, prognosis and lives.

The concept of the Smart Grid is summarised as "encapsulating the entire power generation and distribution system in a single frame" [76]. It makes the whole system cleaner and smarter through intelligent control systems, renewable energy, smart switchboard and other facilities. Supervisory control and data acquisition (SCADA), energy management system, grid communication system and distributed energy resources (DER) are the four components of a smart grid [80]. Among them, SCADA is often attacked because it contains many user data. In 2010, a SCADA system at a nuclear facility in Iran was targeted [81]. There are also data security issues and user privacy in smart grid systems. In 2015, three electricity distribution companies in Ukraine were hacked, causing 225,000 customers to lose power for over three hours [82].

IoV – also called intelligent transportation or connected vehicles [76,83] – helps the transportation system improve road safety, space utilisation, traffic congestion, control costs, and reduce environmental impact by integrating road and vehicle data. By adopting this technology, smart factories realise optimised management of raw material transportation and the Internet of Things. The IoV network contains vehicle-to-sensor, vehicle-to-infrastructure, vehicle-to-pedestrian, vehicle-to-vehicle, and vehicle-to-network communications

implemented via cellular, Bluetooth, and Wi-Fi. Through the seamless integration of these communications, each vehicle can be helped to plan the optimal route. However, some cyberattacks can steal the way by modifying data [84]. Fast network authentication frameworks are currently used to defend against attacks [76].

The Industrial IoT (IIoT) and smart manufacturing are closely related. Smart manufacturing focuses on the manufacture of products, while IIoT refers to the exchange of data between machines, systems and people [85]. The Industrial Internet of Things is the basis for realising smart manufacturing. Due to the industry reliability, security and timeliness requirements, many wired transmissions are often involved in the Industrial Internet of Things [74,75,86]. This is mainly because industrial scenarios are often complex, so there is a lot of interference, and wireless transmission cannot meet the demand. 5G is seen as a solution for the Industrial Internet of Things. However, it lacks licensed frequency bands and business models [87]. EC-GSM-IoT and LTE Cat M1 (LTE-Advanced Pro) are two industrial 5G cellular spectrums currently under development [85]. Bluetooth Low Energy (BLE) is a low-power solution with disadvantages such as high latency, volume and distance limitations [88]. Some researchers solve these problems using connectionless schemes and optimising output distribution [89,90]. The industrial Internet of Things has four main challenges: energy efficiency, real-time performance, limited spectrum, and security [85]. Many IIoT devices run on batteries, so effective energy-efficient designs are necessary. Achieving real-time response in an industrial environment with multiple disturbances is a must for IIoT. The channel congestion and resource allocation problems caused by using many wireless devices in the limited spectrum are also difficulties in industrial Internet design. Data security and privacy issues are also some elements that must be considered in the design. 6G is considered one of the effective solutions to the difficulties of the Industrial Internet of Things, but it is still in the research stage [91,92].

The Internet of Everything (IoE) concept is built based on IoT, which connects people, data, processes, and things rather than just focusing on things [86]. This concept was first proposed by CISCO in 2012. IoE focuses more on intelligent network connections and technologies, while IoT focuses on physical devices and network infrastructure [93]. IoE has received more attention in Industry 5.0 because the core of Industry 5.0 is people, and IoE pays more attention to the connection between people and things. The connection that IoE focuses on is realised through HMI. Therefore, IoE and HMI are inseparable from each other.

The core of IoE is the process [86]. The entire system can become a whole through the flow of people, things, and data. It can analyse and extract the collected information in real-time and provide more precise assistance and improvement without IoT's environmental interference [87]. The development of IoE faces challenges similar to those of IoT. However, the difference is that since IoE also connects people, it requires people to have a consensus on the entire system. This is the basis for realising IoE.

3.3.2. Cloud computing and edge computing

The United States National Institute for Standards and Technology (NIST) defines Cloud Computing (CC) as "a model for enabling ubiquitous, convenient, on-demand network access to a shared pool of configurable computing resources (e.g., networks, servers, storage, applications and services) that can be rapidly provisioned and released with minimal management effort or service provider interaction" [94]. This technology is not new. Since 2000, commercial CC services have appeared on the market [95]. This technology enables remote management and sensor miniaturisation and is an indispensable part of HMI in smart factories. After years of development, it has become a mature product and has been applied in the industry to a certain extent. CC models are mainly divided into four categories [19,95,96]:

• Software as a Service (SaaS): Service providers provide software and applications to users.

• Platform as a service (PaaS): The service provider provides a platform on which users can use their own software and the provided computing power.

• Infrastructure as a Service (IaaS): Users can control all the rented software and hardware anytime and anywhere.

• Container as a Service (CaaS): Similar to PaaS, the applications developed by users are independent of the platform.

Customers can choose suitable model(s) for use according to different needs. However, although CC already has mature models, it still has one of the biggest problems: data security. The statistic [96] shows that in the CC field, cloud security has seven main problems:

- Data model IaaS concerns
- · Data synchronisation issues
- · Secured communication between two parties
- · Data confidentiality and availability
- Data storage security concerns
- Data intrusion threats
- · Data tampering and leakage

Among them, data tampering and leakage problems concern customers the most [90]. Some cloud service providers have recently introduced blockchain technology into the cloud service system to reduce the probability of hacker attacks [97,98]. But this inevitably leads to another problem: the timeliness of data. To reduce latency, many manufacturers need to redesign sensor systems within the factory to ensure the lowest physical latency [96]. Even so, for some tasks with very high timeliness requirements, CC still cannot meet their needs. Therefore, EC was introduced to solve the data tampering and leakage problem.

EC is the opposite of CC. Its main goal is to disperse computing to the network's edge, thereby solving the delay problem caused by transmission [74]. This method can reduce latency and data channel load, effectively reduce energy consumption and ensure data security [4]. EC is usually located between the cloud and end devices and consists of five parts [74,99,100]:

- Authentication and Authorisation: Determine system control rules.
- Offloading Management: Design the offloading scheme and define the information type.
- · Location Services: Map physical locations.
- System Monitor: Provide information such as workload and usage to other parts.
- · Resource Management: Responsible for resource allocation.
- · VM Scheduling: Design the optimal strategy.

EC has the characteristics of wide distribution, relatively small capacity, low energy consumption, low cost, and low delay, which are determined by its nature [101]. EC has been widely used in smart grids, smart cities, smart homes and other fields [74]. In IoT, EC also plays a very important role. For example, facial recognition and motion detection algorithms can be implemented through EC [68,96]. Although EC avoids leakage during data transmission, it still has data security problems. This is because devices at the edge often do not have sufficient capabilities to perform complete security verification [74, 102]. Therefore, both physical and virtual measures must be used to ensure information security.

3.3.3. Machine learning and deep learning

The analysis of big data in the industry is inseparable from machine learning and deep learning algorithms. This is because the data in the manufacturing scenario is large and has many characteristics that correlate with each other [19]. They are also the cornerstone of data analysis in CC and EC. The application of conventional machine learning algorithms in areas such as predictive maintenance is not new [103]. Deep learning has also begun to be applied in manufacturing scenarios in recent years. However, due to its "black box" (the specific operations and processes performed in the model cannot be described) characteristics, it cannot fully meet the industry's safety and controllability requirements. As a result, the use of deep learning in many industrial scenarios is limited [5,104].

Machine learning can be divided into three categories: unsupervised learning (discovering information from data), supervised learning (given data, predicting the label of the data), and reinforcement learning (given data, choosing actions to maximise long-term reward). Research on reinforcement learning has received much attention because it can continuously optimise the model and give more accurate results. Commonly used unsupervised learning algorithms in industry include SVM/SVR, Bayesian Networks, Naive Bayesian Networks, K-nearest Neighbours, Artificial Neural Networks, Multiple Linear Regression, Decision Tree/ Regression Tree, Addictive Models, Logistic Regression, Bag of Words, Locally Weighted Training. Supervised learning algorithms include Kmeans and Self Organising Maps, and reinforcement learning algorithms include PILCO and SMART [104].

Generally, deep learning is divided into unsupervised and supervised learning as a subcategory of machine learning. Commonly used supervised learning algorithms in the industry include Convolutional Neural Networks, Recurrent Neural Networks, Restricted Boltzmann Machines, Multiple Linear Perceptron, and YOLOv2. Unsupervised learning algorithms include Auto Encoders, Convolutional Neural Networks, Recurrent Neural Networks, Restricted Boltzmann Machines, and CAMP-BD [104,105]. Deep learning has been used in image recognition, anomaly detection, natural language processing, and high-dimensional data processing. Compared to machine learning, deep learning is often used on large amounts of data. Applying deep learning to small data sets can result in high variance and low accuracy. Deep learning has imposed higher requirements on hardware, and at the same time, the computational cost is relatively higher.

In recent years, with the continuous improvement of machine learning and deep learning algorithms, the manufacturing industry has seen an increasing application of them. Machine learning and deep learning in manufacturing have three main development directions: task, technology, and industry-centred [106]. Task-centred machine learning research focuses on using algorithms to solve specific tasks, such as defect detection in additive manufacturing [107], process improvement [108], and optimising production processes [109]. Technology-centred research focuses on specific technologies that integrate machine learning algorithms, such as processing monitoring systems [110], using soft computing to improve processing performance [111], and using image recognition to monitor production processes [112]. In addition, the transformation of industrial data to big data is also a research focus in this direction [113]. Industry-centred research focuses on how machine learning affects manufacturing development from various perspectives, such as Industry 4.0, Industry 5.0, and smart manufacturing. New technologies such as augmented reality, data mining, the Internet of Things, intelligent supply chain management, cyber-physical systems, intelligent robots, smart factories, virtual manufacturing, and cloud systems are inseparable from the development of machine learning [114]. In addition, deep reinforcement learning, deep learning combined with big data digital transformation, and industrial artificial intelligence are also research focuses that have received much attention [115]. Deep reinforcement learning has the principle of trial and error and can continuously optimise itself in interacting with the environment. Therefore, it has self-adaptation advantages, strong flexibility, and high time efficiency, which can better meet smart manufacturing requirements [116]. The application of this technology in the manufacturing industry is still lacking in sufficient research [117]. The deep integration of deep learning and big data technology provides new technical support for the digital transformation of factories. Deep learning ensures factories can make more intelligent and sustainable decisions and deployments during digital transformation [118]. Industrial artificial intelligence will become the brain of future smart factories, making decisions on the deployment and arrangement of factories [115].

In future smart manufacturing scenarios, a major change is expected to occur. Human and environment-related data will become one of the focus areas of data analysis to optimise the HMI experience. These two types of data are areas where deep learning is suitable. Their requirements for timeliness are not so high, so even in an environment with low computing power and high latency, deep learning can be used to offer a suitable response. In addition, deep learning has shown great potential in human-related fields such as social media and medical practice, so it has become promising as an effective tool for assisting people in industrial contexts. To meet the needs of individualisation in the future, factories need to adopt algorithms with self-optimisation effects to improve the factory's ability to manufacture different products.

3.4. Data transmission mechanism

Data transmission sends and receives data and signals between sensors and data processing centres. The data transmission mechanism in the manufacturing field can be divided into wired "Industrial Ethernet" and wireless WSN according to the connection method [19]. In addition, according to different wireless spectrums, the transmission mechanism can also be divided into 5G, 6G, etc. [91]. Different mechanisms have different properties and are suitable for different industrial environments. In an industrial environment, the security issue during data transmission in HMI is one of the most important issues [74,96-98,102]. In WSN, many studies have pointed out security issues due to wireless transmission [119-121]. In addition, because industrial scenarios are often complex and have multiple interfering signals, the use of WSNs is also limited [19]. "Industrial Ethernet" is currently the world's most successful and widely used wired communication architecture [103]. After years of development, Industrial Ethernet has the advantages of fast transmission speed, high adjustability of network scale, and convenient parallel connection with other networks. But it also has disadvantages. These data transfer devices were not designed for the harsh environment of a factory, so the extra expense was spent on protective enclosures and add-on equipment to keep them functioning properly. Second, it has an inherent non-deterministic nature, which makes it unable to meet the time requirements of some industrial equipment and models [122].

WSN will be one of the most important enabling technologies in Industry 5.0 because it can better meet the sustainability and humancentric demands of Industry 5.0. In addition, 6G will also help WSN to accomplish these goals better as it can efficiently improve the data transmission speed and coverage. Therefore, this section will mainly introduce these two technologies, the challenges they face and the opportunities they bring.

3.4.1. Wireless sensor network

WSN refers to a series of sensor nodes dispersed in space and connected by wireless transmission [123]. These sensor nodes measure fluctuations in the surrounding environment, process the information, and send it to designated devices [124]. Through the cooperation of multiple sensors, WSNs can monitor a wide range of counterattacks. In recent years, with the continuous development of technology, these wireless sensors have achieved characteristics such as miniaturisation, low cost, and high sensitivity, so they have expanded from the military field to industrial, medical, maritime exploration, environmental monitoring and other fields [125].

WSN consists of four parts: WSN hardware, WSN communication stack, WSN Middleware, and Secure Data aggregation [75]. WSN hardware includes a power supply, transceiver, processor, and sensor interface. These hardware generally consist of multiple A/D converters that can communicate in the same frequency band, ensuring the network's versatility. The WSN communication stack utilises topology, routing and MAC layer to realise the connection between the WSN subnet and the Internet. WSN Middleware is a platform-independent middleware used to isolate and access resources between different sensors. The main goal of Secure Data aggregation is to ensure the system's security and provide self-healing capabilities.

The basic feature of WSN is intra-network transmission, which does not transmit original data but integrates them instead [126]. In recent decades, WSN has played an important role in supply chain management, manufacturing process control and automation [127,128]. By integrating WSN and existing networks, smart factories reduce the labour force, improve automation levels, and increase revenue [120]. WSN is also widely used in wearable sensors, which promotes the development of HMI [2]. Whether it is a wearable device based on VR or a smart glove that remotely controls a robot, WSN is indispensable [26,32,64]. However, as mentioned above, there are still some difficulties and challenges in applying WSNs. The biggest concerns are those about data and security [120]. With the explosive growth of the total amount of data in the industry, it is difficult for factories to manage such a large amount of data in a timely manner [129]. Difficulties with system updates also contribute to the problem. Because these devices are often distributed across various factory parts, the amount of work and disruption needed to upgrade the equipment is dramatic. This leads to a relatively weak anti-attack negligence of the whole system. Sensors deployed in some vulnerable environments become weak points of the overall system. Also, the risk of being attacked is aggravated because the supply chain system is often connected to the external network or cloud system [130]. Finally, the limited resources of WSN make it unable to respond to changes in external conditions effectively. This resource limitation is reflected in two aspects. The energy limitation makes the sensor unable to switch between working and sleeping modes flexibly, and the Internet integration is not high enough due to the restricted movement [131]. Researchers hope to solve the current problems of WSNs by deploying 6G in the future. Therefore, developing WSN devices with sufficient computing power, low energy consumption and low cost in 6G is also a major challenge in Industry 5.0.

3.4.2. 5G/6G

5G has become one of the most popular technologies in recent years. It extends the connection between people to people and things so that popularising a series of technologies requiring high data transmission speed, including unmanned vehicles, smart medical care, autonomous manufacturing, and remote operations, is no longer restricted [132]. Compared with 4G, 5G has been significantly improved in many aspects. However, it still has shortcomings such as power consumption, secure connection, latency, dense connection, and global coverage. Therefore, academia and industry have begun to focus on 6G research. The main goal of 5G is to realise the "Internet of Everything" (IoE). And 6G hopes to achieve intelligent IoE by integrating AI technology. The biggest difference between 6G and 5G technology is that 6G not only includes communication technology. 6G will combine communication and sensing technologies to enable a series of functions, such as positioning, sensing, and communication, to be realised through it [133]. [91] summaries the future development of 6G as a six-F trend: full spectrum, full coverage, full dimension, full convergence, full photonics, and full intelligence. Another research [133] described the goal of 6G as "global coverage, all spectrum, full applications, all senses, all digital, and strong security".

6G expects to achieve global coverage communication services by integrating various communication networks (e.g. unmanned aerial vehicle (UAV) communication, ground ultra-dense communication network, maritime communication, satellite communication, and underground communication). This service will incorporate remote areas and special scenarios into the communication system [132]. Full spectrum can not only solve the problem of band spectrum congestion caused by the explosive growth of the number of devices but also provide higher data transmission rates. Sub-6 GHz, centimetre wave (cmWave), mmWave, THz [134], and optical wireless bands [135] will all be used in 6G [133], 6G's ultra-short-distance communication, wireless data centre, and nano-Internet cannot be realised without the support of THz's ultra-high bandwidth and ultra-fast transmission rate. Optical wireless bands have the characteristics of energy saving, high security, no electromagnetic radiation, and no frequency band, etc., and can play a huge role in 6G. Full applications aim to combine 6G and robotics, AI, big data and other technologies to realise intelligent applications such as smart cities and IoE and provide human beings with personalised and diversified services. The development of all senses focuses on combining 6G with wearable devices and VR, integrating go-you and display, and providing people with a full range of sensory experiences. This will effectively promote the development of remote surgery, skills training and other fields [136]. All digital will further improve the realtime and accuracy of digital twin technology through 6G to realise the deep integration of physical and virtual. Communication security is a hot spot in 6G research. Current research attempts to combine security technologies such as quantum communication [137], blockchain [138], and 6G to ensure data security during data transmission. Many standardisation organisations, such as IEEE and ITUT, have proposed 6G system security architecture and process standards [139].

In HCSM, 6G will be one of the most important enabling technologies. It can bring ultra-low latency and ultra-high-speed data transmission services to smart factories, providing wider coverage and more diversified and customised services [92]. The user experience can thus be improved during the HMI process. The tactile internet is considered an important future application [140]. In industrial settings, telerobotics will act as human avatars. The user's tactile feedback will be transmitted to the robot in real-time through the network, and the robot will be controlled to complete various dangerous tasks. Wireless brain–computer interface, immersive XR, and holographic communication are all inseparable from the superb transmission capabilities of 6G [133,141]. These hardware and technologies form the basis of HMI, and interaction and collaboration behaviours can be completed. The next section will focus on topics that attract researchers' attention in interaction and collaboration.

3.5. Interaction and collaboration

In the research of the HMI field, human–machine function and task allocation, trust and workload allocation are three major issues researchers focus on. Moreover, these issues are expected to continue to receive attention as the degree and scope of automation increase.

3.5.1. human-machine function and task allocation

The distribution of functions and tasks between humans and machines has received continuous attention in human-machine interaction [10]. For example, one of the goals of introducing an automated system is to reduce the workload of humans by using machines to complete or assist with some tasks. However, in different HMI designs, the setting of function allocation is different because researchers have different understandings of the functions that humans and machines are good at. The functions in a smart factory consist of four parts: operation and supervision, management, maintenance services, and cybersecurity [2]. The realisation of these functions is mainly through workers performing corresponding tasks. To achieve operation and supervision functions, workers must utilise HMI to monitor machines and processes, control production parameters, identify and assemble parts, and schedule machines. The tasks involved in the management function include administration, workforce and operation scheduling, and production planning. Maintenance services involve fault detection, repair, quality inspection, and simulation. Tasks involved in network security include data retrieval, authentication, security assessment, and network configuration. As automation increases, some functions and corresponding tasks have been replaced by machines, while some still require human participation. Therefore, in a dynamic production environment, workers should not only focus on the tasks they are currently performing but also understand the functions they want to achieve and the shortcomings of the machine. Humans are generally considered better suited for tasks like creativity and decision-making, while machines are better suited for repetitive tasks [142]. However, this does not mean that all repetitive tasks can be completely separated from operator monitoring. Therefore, the operator can misunderstand what task has been replaced by the machine and what task he should focus on.

A good example is the "irony of automation": Introducing automation equipment into the system changes human behaviour [11]. The goal of introducing automation is to help or even replace humans for some work tasks so that humans can focus more on others — and/or free up resources along the way. Still, these automation devices can also reduce people's concentration in these non-automated tasks and lead to impoverished cognitive processing of information, which impacts cognitive factors such as memory and problem-solving. Studies of driver assistance systems have shown that drivers become less attentive to tasks such as steering, which the system assists, and lose situational awareness [10,143].

The distribution of functions and tasks in HCSM factories will face two major problems. First, the mental load on workers will increase significantly as the professional level of factory equipment increases. The introduction of automation has effectively reduced the physical burden on workers, but the consequence is that a large amount of information is collected. Workers need to process the collected data and information to make correct decisions. Although AI and big data analysis can help people process this data because people are the core and decision-making layer of the factory, the mental load on operators is still heavy. In addition, the growth of non-professional users is also a problem that cannot be ignored. It is unrealistic to expect workers to understand the function of every part of the factory. Workers may not understand the mechanics of the equipment they operate. Good guidance is necessary to assist workers in completing their tasks. Therefore, HMI design in Industry 5.0 should include two considerations: to help professionals reduce their burden and to help non-professionals understand the tasks.

3.5.2. Incorrect use, trust and confusion

Incorrect use, trust and confusion are three topics often discussed together [10]. There are four different kinds of use in HMI: use, misuse (over-dependence and over-trust), disuse (do not use, do not trust) and abuse (do not consider the impact of introducing machines, "irony of automation") [14,19]. Different initial trust levels in the system will lead to different users' use of the device, and the consequences of incorrect use may further affect users' trust in the system. Incorrect use will not only affect human–machine collaboration, reduce efficiency, and cause safety issues but also lead to loss of workers' trust in machines.

Confusion is mainly produced by system complexity [10]. As the factory develops towards customisation and integration, the system needs multiple modes to meet different needs. The manufacturing environment often involves a dynamic set of systems, and as time changes, its operating mode will also change. Different operating modes often appear to deal with different situations. But this can potentially cause mode confusion on the operator's part. Research on autonomous driving has found that confusion may arise for three reasons. Unexpected situations such as sudden snowfall and overtaking vehicles can cause the car's mode to change before the user has enough reaction time to respond to the new mode. However, users often must respond quickly when the unexpected happens [144]. In addition, the vehicle's functions will change as the environment changes. Certain weather and environments may limit the functionality of some sensors, making it dangerous if the operator's attention is not on driving [15]. This situation is more common in manufacturing environments. Since the factory's equipment status and production conditions are constantly changing, it can be said that most of the time, the functions of the factory are changing [19]. "Irony of automation" is the third reason users

pay less attention to mode changes. The introduction of automation will reduce operators' attention to the system, preventing them from detecting changes in the system. In future factories, automated equipment will be used in more complex situations, so multiple operating modes are necessary [11]. Research on modelling and analysis of mode confusion in different manufacturing environments will be one of the most efficient methods of evaluating and improving this problem. An early study found that power plants have high requirements for equipment maintenance, so some equipment will always be in maintenance status. In such an environment, workers can easily become confused about alarms caused by accidents and maintenance [144].

Overall, ensuring that users have sufficient trust in the system is one of the main ways to encourage effective and sustained use and prevent misuse and possible disuse. A bigger problem is that because many technologies and products are developed without user trust, these devices may not have established trust models, and it is very difficult to build models for them. Researchers should not only consider establishing operators' trust in equipment and appropriate evaluation models to accurately assess the degree of trust and prevent the occurrence of excessive trust but also focus on restoring and maintaining trust after an accident. The prevention of confusion should focus on the establishment of an early warning system. This is also inseparable from establishing analytical models for specific environments.

3.5.3. Workload allocation

Several papers within the review were focused on workload and how it can be optimally distributed to achieve tasks at hand better [10,19,145-150]. In general, workload distribution aims to prevent underload and overload so that users are optimally engaged in tasks with enough cognitive resources to perform them effectively. Many studies reviewed often also focus on focus and/or divided attention, as well as attention management, and these will also be discussed below [10].

One purpose of automation is to free up workload (resources) so that people do not have to perform as many repetitive tasks and instead can focus on other tasks where human involvement is more valuable. However, this does not mean that these automated tasks do not require human intervention - and hence workload. When failures or abnormal conditions occur, workers must still participate in these tasks to supervise. Numerous studies have found that workers in automated control environments are less sensitive to alarms [10,19,145]. To this end, a study has used early warning to prepare workers for emergencies in advance [146]. In addition, the researchers proposed six solutions to keep workers focused: avoiding the role of human supervision of automation, reducing the role along an objective dimension (reduce automation tasks and devices), reducing the role along a subjective dimension (add human tasks), support the role from the behaviour paradigm (more training), support the role from the dyadic cognitivism paradigm (align content, structure and functions of computerised systems with human minds), support the role from the triadic ecological paradigm (emphasise the importance of direct perception and informed considerations of adaptation to specific work domains) [147].

Years of research have found that supervising automated processes is not where humans excel. Human errors still occur when people supervise automated equipment that is designed to prevent or at least minimise operational errors. Therefore, it is necessary to reasonably evaluate whether a certain function is suitable for introducing automation. The introduction of automatic flying systems in aircraft sometimes fails to improve safety and, in some cases, causes the pilot to be away from tasks for a long period, and this can keep them out of the loop and lead to errors and potential accidents in some (albeit very few) cases accidents [147]. Research in this field is also called "stay tuned". Other researchers have found that although longterm supervision is difficult for operators to achieve (due to vigilance limitations), requiring operators to perform a short operation over a short period can significantly reduce the occurrence of errors [148]. Periodic proactive engagement maintains operator engagement and response times better than reactive handling when an incident occurs. For the system, sometimes the operator can be made more involved in the system by increasing the operator's workload, although this seems counter-intuitive. One study found that autonomous driving systems can attract the driver's attention by showing a lack of ability or even deliberately making wrong judgements [149]. A major focus in future operator training should be creating immersive applications that can motivate operators to operate correctly. Some VR operation training equipment is designed based on increased engagement. During training, workers build a better understanding and trust in the system and develop their interest in work. To ensure that computerised systems are consistent with human thinking, it is necessary to explore workers' mental models and follow the user-centred cognitive interface design concept in the design of HMI. The part of the system displayed to the user must meet the four key points: the process is understandable, the results are understandable, consistent with user goals, and meets user reliability expectations [150].

Finally, external environmental context and task considerations should not be ignored in the design of HMI. The operators always perceive the external environment, which directly affects their behaviour. While considering people's psychological state, constructing a suitable external environment and setting appropriately challenging tasks can improve operators' feelings and ability to cope with the workload.

In HCSM, research needs to focus more on how to make worker and machine work partners (and to optimise this) rather than being studied separately as two individuals. The changes brought about by the development of AI to HMI may be inflammatory. These AI-based interactive systems can improve people's communication experience in a humanised way and stimulate recognised creativity and thinking ability in more creative tasks like an assistant, improving workers' comfort and satisfaction. This may be one of the best ways to solve these HMI problems.

In a manufacturing environment, sensors are the first hardware devices to interact when humans and machines interact. They collect signals and information, convert them into data, and pass them to the data centre for analysis. Various sensors based on different mechanisms and with different performances build the foundation for big data analysis. The application of CC and EC in different scenarios allows factories to solve the problem of insufficient data processing and/or transmission capabilities to a certain extent, thereby helping factories realise IoT/IoE. Machine learning and deep learning algorithms have become important means for factory data analysis. In the future, with the popularisation of 6G, WSNs and networks in factories will have stronger anti-interference capabilities and data transmission capabilities, improving the factory's data analysis and collection capabilities and data security. Finally, as the concept of human-centric gradually becomes popular, more and more products and equipment will consider human factors during production and design, which will play an important role in the construction of HMI analysis in these scenarios in the future. In Industry 5.0, machines and humans should cooperate and coexist, trust each other, allocate tasks reasonably, and achieve the goal of improving human well-being.

4. Towards human-centric smart manufacturing: Challenges and opportunities

The previous section reviewed some important research and enabling technologies in the field of HMI, thereby providing some basic understanding for analysing how they will realise the Industry 5.0 vision. This section will focus on the challenges faced when realising the vision of Industry 5.0 and the opportunities that Industry 5.0 brings to future manufacturing.

4.1. Challenges

4.1.1. Non-deployed technology and costs

Applying new technologies to HMI in real manufacturing scenarios can be challenging. Although some developing technologies such as ChatGPT and AI-based voice recognition assistants have been widely used in people's daily lives, technologies such as brain–computer interface and 6G are still mainly based on laboratory research. The development and application of these technologies mainly have three constraints. To design a suitable HMI product/system, factors such as material selection, ergonomic design, equipment performance, etc., must be considered in terms of hardware. At the theoretical level, it is also necessary to consider how humans and machines collaborate, how work is distributed, how to ensure workers' trust in machines, and how to evaluate and monitor these indicators quantitatively. The complexity brought to HMI system design involving multiple disciplines and fields is a major restriction for HMI technology to be applied in actual scenarios [10].

In addition, cost is also a major factor restricting the development of these technologies. Big companies with high-profit margins, such as Intel, Nvidia, Tesla, etc., are the main drivers of HMI research in manufacturing [16]. However, those companies with low-profit margins need more incentives and strength to invest in this area, even if these studies can bring huge benefits to the company in the long run. Manufacturers with low-profit margins often find implementing measures and equipment with long payback periods difficult because their capital pools are not generous. Therefore, Industry 5.0 should focus on how to reduce the cost of HMI technology and deployment.

Sustainability has been discussed since the age of Industry 4.0. But it has now turned into an urgent need. The realisation of sustainability cannot be achieved without applying new technologies. In recent years, incidents and events such as the COVID-19 pandemic and the Russia-Ukraine war have exposed the fragility of existing international supply systems. Global warming is also making sustainability an urgent need in industry. Countries have passed bills to stop using highly polluting gasoline cars in the next two decades [8,9,19]. To implement the concept of sustainability in HMI in the industry, factories must adopt greener equipment and sensors from the technical level and improve workers' understanding of this concept. Improving the explainability of the system will not only prevent the emergence of unnecessary processes but also encourage workers to implement appropriate behaviours to improve the sustainability of the factory [130,131].

4.1.2. The application of human-centric concept: from efficiency and profit to human-centric

Although Industry 5.0 have proposed the human-centric concept, it is difficult to expect enterprises to focus more on human than their profit-pursuing nature. As discussed in previous sections, improving factory output through people's well-being is a long-term goal. Most small and medium-sized enterprises lack the corresponding funds and sometimes the motivation to undertake related investment and industrial upgrading. An important point of contradiction is that low-end manufacturers are where people's well-being is neglected the most. Still, they lack the corresponding funds to improve the system in this area because their demand for innovation is relatively low. Therefore, implementing Industry 5.0 not only depends on enterprises but also needs help from the government and policy.

In addition, for factories, the human-centric concept should be implemented at the beginning of design to fully understand and meet human needs. In a human-centred factory, everything should be about serving people. Devices and systems based on AI should continuously enhance their collaboration and empathy capabilities based on realtime data and help workers maintain a good physical and mental state at all times.

A comprehensive HCSM factory should include human-centric process planning, workshop scheduling, processing and assembly [151]. Human-centric process planning requires factories to consider workers' needs when coordinating production processes. Some researchers have explored the differences in decision models under different design philosophies (e.g. [152,153]). Human-centric workshop scheduling focuses on putting the interests of operators and consumers in the first place in production in the workshop. AI can help factories complete intelligent workshop scheduling, thereby assisting operators and improving efficiency [154]. Human-centric processing focuses on recognised safety and health. Remote operations and more humanfriendly smart assistants are technologies already used in some smart factories [7,26,32,64]. Human-centric assembly is mainly based on the following considerations. First, humans still dominate the assembly process. Second, assembly work is repetitive. Therefore, this part mainly aims to use automated equipment to replace repetitive tasks and fully stimulate the operator's cognition and flexible innovation capabilities.

Besides, research in the HMI field in the industry has yet to focus on human-related data. Human-related data is expected to be central in HCSM, as it is one of the indispensable data to help HMI devices understand people. Collecting and analysing these data allows the machine to provide humanised and personalised assistance to the operator. This makes the IoE that the industry has always imagined cannot be well implemented. More effort must be put into human-related data in deploying and applying sensors, which also puts forward new requirements for developing technology and research [9].

4.1.3. Training of employees

Training employees is also a big challenge that needs to be optimised to reap HCSM's rewards fully. These training efforts should train workers to understand the equipment and teach them how to use it as a supporting feature of their work lives [10].

In some previous sub-sections, we explored the growth of nonprofessional users and the emergence of professional users as the core of manufacturing. It needs to be clear that with the increase in integration and complexity of manufacturing systems, even professional users need help understanding the entire system. In today's fast-paced technological iteration, operators in manufacturing scenarios have both the attributes of professional users and the attributes of non-professional users. Therefore, a major difficulty in worker training is that different workers have different knowledge backgrounds, and it is difficult to achieve the desired effect only by centralised training [155].

Therefore, personalised training based on big data and AI is necessary. This personalised training should focus on helping workers understand the system rather than on subtle technical difficulties. At its core, HCSM is about helping workers unleash their creativity and flexibility while safeguarding their well-being. Training that sticks to the HCSM core can help workers better understand factory systems and make the best decisions.

4.1.4. Data and personnel security

Security is also a major issue that factories will pay attention to. Security can be divided into two issues: data security and personnel security. Data security not only depends on confidentiality technology but also on workers' awareness of protection. In the HMI process, how to transmit this concept to workers will be a more permanent and effective way to improve. The safety of workers is inseparable from the addition of humanised design, such as the empathy ability of the interactive system. These designs improve worker comfort and give them a greater awareness of hazards.

In recent years, blockchain technology has also been applied to data encryption in the IoT field due to its confidentiality. Blockchain has the characteristics of traceability, anti-tampering, and decentralisation, so it has received much attention in ensuring information security in industrial IoT [91,92]. However, this technology still faces three major problems [151]:

- The blockchain model relies on the IoT architecture, and its ability to deal with problems alone is weak, and its ability to deal with physical attacks is weak.
- The current technology is immature, and there still needs to be more research on how to evaluate the security and privacy of the network.
- How to promote the integration of people and systems and how to protect the privacy and security of people still needs to be solved.

In addition to ensuring physical safety, factories should also ensure workers' psychological safety and privacy. The manufacturing industry has relatively mature processes and measures to protect workers' health and life safety. Considering psychological conditions and privacy protection will receive more attention in Industry 5.0. Since humanrelated data will become an important part of HCSM data, these data are closely related to workers' psychological state and privacy. Due to their natural attributes, technologies such as brain–computer interface and myoelectric-based myoelectric interaction must consider how not to infringe on human privacy when designing. The goal of Industry 5.0 is to enable humans and machines to work closely together rather than manipulating workers through HMIs. Therefore, how to protect workers' mental health without abusing these data and models will be a problem that needs to be addressed in future research.

4.1.5. Real-world HMI implementation

To ensure that HMI systems can be deployed in the real world, some principles and factors must be considered during the design process of HMI systems. As research continues to deepen, people have established different design principles for HMI systems [156]. The frameworks of Shneiderman [157], Wickens, Lee, Liu, and Becker [158] et al. believe that user experience, needs, and capabilities should be the core of the design. The principles summarised by Shneiderman include system consistency (using consistent instructions), providing feedback on input (sending status information when the system has a request), designing a closed loop of dialogue (task completion prompts), providing simple error prompts (avoiding multiple clicks), and reducing cognitive memory load (avoiding complex menus). Wickens et al. are more concerned with human cognitive abilities and limitations and therefore regard perception (clear interactive interface), attention (reducing the difficulty of obtaining information), memory (simplifying the steps to obtain important information), mental models (ensuring that icons and logic are easy to understand), and situational awareness (the ability to predict the future and remind of key past events) as general HMI design principles [158]. Cuevas [159] proposed a four-factor criterion: ethnographic/anthropomorphic traits, cognitive factors, predictive modelling, and empirical testing. This criterion states that HMI systems must be evaluated from these four perspectives to meet practical application requirements.

During the design process, we believe an HMI system must be verified to meet the requirements from the perspectives of adaptability, accessibility, usability and functionality. Adaptability is the ability of the HMI system to adapt to changes in the environment, tasks and personnel. A good HMI system should be customised according to user needs and the external environment to ensure user experience. Accessibility refers to whether the HMI system is easy to use. In particular, when users have special needs or physical disabilities, the HMI system should ensure they can still access it. Usability refers to the efficiency and ease of use of the HMI. A good HMI system should be easy to understand and reduce the user's learning cost and error probability. Functionality means that when designing the HMI, the system should ensure the required functions are available to prevent functional deficiencies.

4.2. Opportunities

Although implementing Industry 5.0 faces many challenges, it will also bring new opportunities to the HMI system in the manufacturing scenario. Next, three emerging technologies will be introduced. They are expected to drive the development of HMI in Industry 5.0.

4.2.1. Digital twins and industrial metaverse

Digital twin (DT) is a technology that digitises physical devices and processes to enable remote manipulation, prediction, optimisation, simulation, and control of manufacturing processes [160]. It is a discipline that is still developing. Early DT models included three dimensions: physical, virtual, and connected. In this model, models in the virtual space are mapped to the physical world through data transmission [161]. Fei Tao et al. proposed a five-dimensional model that includes virtual models, physical entities, connections, digital twin data, and services [162]. The theory of DT includes four parts: modelling and simulation, data fusion, interaction and collaboration, and services. According to the five-dimensional model theory, DT modelling includes physical modelling (extracting physical features), virtual modelling (building virtual models), connection modelling (building connections between models), data modelling (storing and processing data), and service modelling (identifying and analysing services). Data fusion involves data preprocessing, mining, and optimisation. It helps to build and connect models. Interaction and collaboration include physical-physical, physical-virtual, and virtual-virtual interactions. Physical-physical interaction allows multiple devices to work together. Physical-virtual interaction can connect remote models and entities to achieve manipulation. Virtual-virtual interaction can share data within the network to achieve remote collaboration. Services include business encapsulation, service optimisation and matching, quality assessment, etc., which can help DT provide the best service to customers. DT is used in product design, production process monitoring and health management [160].

The Industrial Metaverse has continued to develop in the manufacturing industry in recent years. It is a concept very similar to Digital twins. Digital twins start from the virtualised design and development of complex products, while the industrial metaverse originated from the game industry and gradually developed into the industrial world. Digital twins are first oriented towards technology, while the Industrial Metaverse focuses more on people [163]. The Industrial Metaverse aims to merge the physical and digital worlds. It builds an online virtual factory parallel to the real factory so that all relevant researchers can work together in the same scenario [164]. In this virtual factory, users can simulate real work scenarios to train, design, test and operate realworld equipment [165]. The Industrial Metaverse focuses on building a complete 3D environment, allowing workers to use and analyse data from the spatial dimension to solve previously unsolvable problems. In the industrial virtual world, when technical experts join an invitation link to a remote factory, they can use wearable devices to see the manufacturing scene visually.

The Industrial Metaverse can bring many benefits to manufacturers. First, it can be optimised and designed through simulation, reducing actual production costs and trial and error costs [166]. In addition, it can also improve the production efficiency of manufacturing companies, allow employees to become familiar with operations through virtual training, and reduce failures in actual operations. The industrial metaverse can promote company innovation. Rapid design, testing, and development of new products through virtual environments can help companies form competitive advantages [167]. The Industrial Metaverse can also help companies become more sustainable. It can maximise resource utilisation while meeting individual needs through intelligent, planned production, procurement, quality and inventory management [163,166,167].

4.2.2. Worker safety, health monitoring and ergonomics design

Sensors and monitoring systems in Industry 5.0 will focus not only on processes but also on the status of workers. Research on worker fatigue in manufacturing environments is not new [168–170]. However, because people's definition of fatigue is relatively subjective, and related analysis requires collecting some data that may involve privacy, large-scale applications of this model have not yet been seen in actual production. With the collection and research of human-related data, future smart factories will have personalised safety and health detection systems for each worker to ensure the well-being of workers [171]. The expected goal is that in the future, big data analytics can be used to create personalised profiles based on workers' different indicators, thereby improving worker well-being from an ergonomic, medical and psychological perspective. Since every worker's situation differs, using universal data to meet their needs is unrealistic. Only a personalised, customised system can meet the needs of each worker.

Ergonomics will also be a big focus in Industry 5.0. Monitoring systems can only provide us with early warnings and indications, while ergonomics will improve conditions that may arise for workers [172]. In the future, in Industry 5.0, a large number of ergonomic equipment will be designed using human-centric methodology [173]. These devices can reduce the cognitive load on workers, reduce the probability of errors, and increase productivity. In addition to this, these devices prevent the occurrence of many occupational diseases and health problems. These devices will also serve as a monitoring system to safeguard workers' well-being.

4.2.3. Brain-Computer Interface (BCI)

Brain-computer interface technology will be an extremely important key technology that will promote social development in the future. This system can connect the human brain and the external environment, allowing users to control devices through brain signals [174]. Researchers have always favoured this field. The system can be classified as passive or active depending on how the brain is used. Passive BCI analyses the brain's unconscious signals and emotions, while active BCI involves the user's voluntary brain movements. Passive BCI can be used in fault and fatigue detection. For example, an early warning is given when a person is exhausted [175]. Active BCI can help users complete interactions with devices.

BCI has broad application prospects. However, most current research on BCI still focuses on the medical aspect [174]. It replaces the central nervous system and controls prosthetic limbs [176]. BCIs are used in some clinical settings to assess and diagnose patients for various diseases [177]. In the industrial field, BCI can effectively promote human-machine collaboration and help humans and robots understand each other's intentions. One study used eye blinks as a signal input to complete loading and unloading work through BCI-controlled robots [1]. In the future, as the number of psychological states and input signals that BCI can classify increases, it can be used to control robots to complete more complex tasks. This can significantly reduce the physiological load on workers and prevent possible accidents. The deployment and development of BCI will greatly promote realising the Industry 5.0 vision.

4.2.4. MES systems in Industry 5.0

The concept of the manufacturing execution system (MES) emerged in the 1990s. It builds a top-down system for managing workshops and enterprise resource planning to establish detailed operating plans for enterprises [178]. Through this system, enterprises can manage workshops in real time. In the past three decades, the industry has established 11 functions of MES systems: resource allocation and status management, job scheduling, product unit scheduling, document control, data collection and acquisition, labour management, quality management, process management, maintenance management, product tracking and performance analysis [179]. In the field of MES, there are currently some different solutions on the market. POMSNet Aquila, ABB MES, Siemens Opcenter, GE Digital's Proficy MES, 3DS's DELMIAworks and PLEX's MES/MOM are some of the popular MES solutions and software on the market [180]. These solutions and software have their characteristics in terms of functions and applicable scenarios. Many companies use a combination of different software to meet actual needs. However, the intelligence level of these existing commercial MES systems is not very high and cannot meet the requirements of enterprises for predictive capabilities [180].

There are three frontiers in the research of MES: combining AI and MES systems, DT and augmented reality [180]. By combining AI, MES systems can handle tasks in the system more efficiently, ensure the safety of the working environment, and meet productivity and quality goals [181]. DT can connect the intelligent layer of MES with the physical world to achieve real-time monitoring and management of the workshop [182]. Augmented reality can promote the collaboration and integration of people and MES systems. In Industry 5.0, people will still play an important role in the factory. In the MES system, augmented reality can visualise data to assist operators in better managing the MES system [183,184].

The development of MES systems will promote the realisation of Industry 5.0. It is an effective measure for smart factories to achieve sustainability and elasticity goals. MES systems combined with augmented reality will make the interaction between operators and systems more humane in the future, thereby making the relationship between people and systems closer. This system can ensure production and is an important guarantee for employee well-being. By incorporating employee well-being into production workflow management, factories can become people-centric.

5. Conclusions

This paper gives a systematic review of the development of HMI in the future HCSM and Industry 5.0 and the constraints that may be encountered by analysing and summarising the existing research and technology in the HMI field. According to the signal transfer process in the HMI, this article classifies and summarises these related studies in four parts: sensor, data processing, data transmission mechanism, interaction and collaboration. In the sensor field, the equipment involved in HMI is summarised from the perspectives of optical, acoustic, bionic, tactile and motion. The review on data processing mainly focuses on IoT and IoE, CC, EC, machine learning and deep learning. The data transmission mechanism discusses WSN and 5G/6G. In the interaction and collaboration part, this article focuses on three main issues in HMI field research: 1. human–machine function and task allocation, 2. incorrect use, trust and confusion, and 3. workload allocation.

Reviewing the current HMI Research in the field, this paper explores the challenges and opportunities that HMI may encounter in Industry 5.0. There are four challenges facing the realisation of the Industry 5.0 vision. The development maturity and deployment costs of technologies limit their practical application. The human-centric concept has not been placed in an important position in the design of products and processes. This has resulted in many equipment and links in the factory being naturally incapable of building HMI analysis models. Workers' current knowledge and skill levels cannot cope with the new equipment and processes of Industry 5.0. Security issues brought about by the personalisation of HMI analysis and increased data collected about people are also important challenges facing Industry 5.0.

Many new opportunities will emerge in Industry 5.0. This paper explores three opportunities related to manufacturing. The industrial metaverse will bring a new online operating model to manufacturers, enhance collaboration and innovation, and enable factories to make more informed data-driven decisions. Ergonomics, human factors and many more intelligent safety and health monitoring systems will bring people and machines closer together and ensure the well-being of people in HMI. Brain–computer interface (BCI) will connect people and machines, with a promise of making future machines truly like part of the human body, serving as human assistants to help people complete manufacturing tasks.

CRediT authorship contribution statement

Jialu Yang: Writing – original draft, Methodology, Investigation, Conceptualization. **Ying Liu:** Writing – review & editing, Supervision, Methodology, Conceptualization. **Phillip L. Morgan:** Writing – review & editing, Supervision, Methodology.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

No data was used for the research described in the article.

References

- Jim Davis, Thomas Edgar, Robert Graybill, Prakashan Korambath, Brian Schott, Denise Swink, Jianwu Wang, Jim Wetzel, Smart manufacturing, Annu. Rev. Chem. Biomol. Eng. 6 (2015) 141–160.
- [2] Naveen Kumar, Seul Chan Lee, Human-machine interface in smart factory: A systematic literature review, Technol. Forecast. Soc. Change 174 (2022) 121284.
- [3] Dimitris Mourtzis, John Angelopoulos, Nikos Panopoulos, Digital manufacturing: The evolution of traditional manufacturing toward an automated and interoperable smart manufacturing ecosystem, in: The Digital Supply Chain, Elsevier, 2022, pp. 27–45.
- [4] Boubakr Nour, Spyridon Mastorakis, Abderrahmen Mtibaa, Compute-less networking: Perspectives, challenges, and opportunities, IEEE Netw. 34 (6) (2020) 259–265.
- [5] Junliang Wang, Chuqiao Xu, Jie Zhang, Ray Zhong, Big data analytics for intelligent manufacturing systems: A review, J. Manuf. Syst. 62 (2022) 738–752.
- [6] Birgit Vogel-Heuser, Dieter Hess, Guest editorial industry 4.0- prerequisites and visions, IEEE Trans. Autom. Sci. Eng. 13 (2) (2016) 411–413.
- [7] Mohsen Soori, Behrooz Arezoo, Roza Dastres, Internet of things for smart factories in industry 4.0, a review, Internet Things CyberPhys. Syst. 3 (2023) 192–204.
- [8] Xun Xu, Yuqian Lu, Birgit Vogel-Heuser, Lihui Wang, Industry 4.0 and industry 5.0—inception, conception and perception, J. Manuf. Syst. 61 (2021) 530–535.
- [9] Yuqian Lu, Hao Zheng, Saahil Chand, Wanqing Xia, Zengkun Liu, Xun Xu, Lihui Wang, Zhaojun Qin, Jinsong Bao, Outlook on humancentric manufacturing towards industry 5.0, J. Manuf. Syst. 62 (2022) 612–627.
- [10] Christian P. Janssen, Stella F. Donker, Duncan P. Brumby, Andrew L. Kun, History and future of human-automation interaction, Int. J. Hum.–Comput. Stud. 131 (2019) 99–107.
- [11] Lisanne Bainbridge, Ironies of automation, in: Analysis, Design and Evaluation of Man–Machine Systems, Elsevier, 1983, pp. 129–135.
- [12] Thomas B. Sheridan, Function allocation: algorithm, alchemy or apostasy? Int. J. Hum.-Comput. Stud. 52 (2) (2000) 203–216.
- [13] Mary T. Dzindolet, Scott A. Peterson, Regina A. Pomranky, Linda G. Pierce, Hall P. Beck, The role of trust in automation reliance, Int. J. Hum.–Comput. Stud. 58 (6) (2003) 697–718.
- [14] Raja Parasuraman, Victor Riley, Humans and automation: Use, misuse, disuse, abuse, Hum. Factors 39 (2) (1997) 230–253.
- [15] Ian Y. Noy, David Shinar, William J. Horrey, Automated driving: Safety blind spots, Saf. Sci. 102 (2018) 68–78.
- [16] Christian Krupitzer, Sebastian Mu'ller, Veronika Lesch, Marwin Zu'fle, Janick Edinger, Alexander Lemken, Dominik Sch'afer, Samuel Kounev, Christian Becker, A survey on human machine interaction in industry 4.0, 2020, arXiv preprint arXiv:2002.01025.
- [17] Francesco Longo, Letizia Nicoletti, Antonio Padovano, New perspectives and results for smart operators in industry 4.0: A human-centered approach, Comput. Ind. Eng. 163 (2022) 107824.
- [18] Kaixuan Wang, Lim Wei Yap, Shu Gong, Ren Wang, Stephen Jia Wang, Wenlong Cheng, Nanowire-based soft wearable human-machine interfaces for future virtual and augmented reality applications, Adv. Funct. Mater. 31 (39) (2021) 2008347.
- [19] Jialu Yang, Tianyuan Liu, Ying Liu, Phillip Morgan, Review of humanmachine interaction towards industry 5.0: human-centric smart manufacturing, in: International Design Engineering Technical Conferences and Computers and Information in Engineering Conference, vol. 86212, American Society of Mechanical Engineers, 2022, V002T02A060.
- [20] Qiuhong Ke, Jun Liu, Mohammed Bennamoun, Senjian An, Ferdous Sohel, Farid Boussaid, Computer vision for human-machine interaction, in: Computer Vision for Assistive Healthcare, Elsevier, 2018, pp. 127–145.
- [21] Bojan Bla^{*}zica, The inherent context awareness of natural user interfaces: a case study on multitouch displays, Informatica (Ljubl.) 38 (4) (2014).
- [22] Mohammad Mehdi Moniri, Fabio Andres Espinosa Valcarcel, Dieter Merkel, Daniel Sonntag, Human gaze and focus-of-attention in dual reality humanrobot collaboration, in: 2016 12th International Conference on Intelligent Environments, IE, IEEE, 2016, pp. 238–241.
- [23] Jung-Sing Jwo, Ching-Sheng Lin, Cheng-Hsiung Lee, Smart technology-driven aspects for human-in-the-loop smart manufacturing, Int. J. Adv. Manuf. Technol. 114 (2021) 1741–1752.

- [24] Florian Echtler, Thomas Pototschnig, Gudrun Klinker, An led-based multitouch sensor for lcd screens, in: Proceedings of the Fourth International Conference on Tangible, Embedded, and Embodied Interaction, 2010, pp. 227–230.
- [25] Chihyung Jeon, The virtual flier: The link trainer, flight simulation, and pilot identity, Technol. Cult. (2015) 28–53.
- [26] Yewande M. Akinola, Oluwatoyin C. Agbonifo, Oluwafemi Abimbola Sarumi, Virtual reality as a tool for learning: The past, present and the prospect, J. Appl. Learn. Teach. 3 (2) (2020) 51–58.
- [27] Hui Sun, Shujin Fan, Car sequencing for mixed-model assembly lines with consideration of changeover complexity, J. Manuf. Syst. 46 (2018) 93–102.
- [28] Anna Brolin, Peter Thorvald, Keith Case, Experimental study of cognitive aspects affecting human performance in manual assembly, Prod. Manuf. Res. 5 (1) (2017) 141–163.
- [29] Abhiraj Deshpande, Inki Kim, The effects of augmented reality on improving spatial problem solving for object assembly, Adv. Eng. Inform. 38 (2018) 760–775.
- [30] Zhen Yang, Jinlei Shi, Wenjun Jiang, Yuexin Sui, Yimin Wu, Shu Ma, Chunyan Kang, Hongting Li, Influences of augmented reality assistance on performance and cognitive loads in different stages of assembly task, Front. Psychol. 10 (2019) 1703.
- [31] James Baumeister, Seung Youb Ssin, Neven AM ElSayed, Jillian Dorrian, David P Webb, James A. Walsh, Timothy M. Simon, Andrew Irlitti, Ross T. Smith, Mark Kohler, et al., Cognitive cost of using augmented reality displays, IEEE Trans. Vis. Comput. Graph. 23 (11) (2017) 2378–2388.
- [32] Fabrizio Lamberti, Federico Manuri, Gianluca Paravati, Giovanni Piumatti, Andrea Sanna, Using semantics to automatically generate speech interfaces for wearable virtual and augmented reality applications, IEEE Trans. Hum.–Mach. Syst. 47 (1) (2016) 152–164.
- [33] Siddharth S Rautaray, Anupam Agrawal, Vision based hand gesture recognition for human computer interaction: a survey, Artif. Intell. Rev. 43 (2015) 1–54.
- [34] Lin Guo, Zongxing Lu, Ligang Yao, Human-machine interaction sensing technology based on hand gesture recognition: A review, IEEE Trans. Hum.–Mach. Syst. 51 (4) (2021) 300–309.
- [35] Jing-Hao Sun, Ting-Ting Ji, Shu-Bin Zhang, Jia-Kui Yang, GuangRong Ji, Research on the hand gesture recognition based on deep learning, in: 2018 12th International Symposium on Antennas, Propagation and EM Theory, ISAPE, IEEE, 2018, pp. 1–4.
- [36] Pramod Kumar Pisharady, Martin Saerbeck, Recent methods and databases in vision-based hand gesture recognition: A review, Comput. Vis. Image Underst. 141 (2015) 152–165.
- [37] Tomoya Murata, Jungpil Shin, Hand gesture and character recognition based on kinect sensor, Int. J. Distrib. Sens. Netw. 10 (7) (2014) 278460.
- [38] Wei Zeng, Cong Wang, Qinghui Wang, Hand gesture recognition using leap motion via deterministic learning, Multimedia Tools Appl. 77 (2018) 28185–28206.
- [39] Yunzhi Ling, Tiance An, Lim Wei Yap, Bowen Zhu, Shu Gong, Wenlong Cheng, Disruptive, soft, wearable sensors, Adv. Mater. 32 (18) (2020) 1904664.
- [40] Yuanjie Xia, Mohamed Khamis, F. Anibal Fernandez, Hadi Heidari, Haider Butt, Zubair Ahmed, Tim Wilkinson, Rami Ghannam, State-ofthe-art in smart contact lenses for human-machine interaction, IEEE Trans. Hum.-Mach. Syst. (2022).
- [41] James B. Kinn, Richard A. Tell, A liquid-crystal contact-lens device for measurement of corneal temperature, IEEE Trans. Biomed. Eng. (5) (1973) 387–388.
- [42] Sarah C. Xu, Angela C. Gauthier, Ji Liu, The application of a contact lens sensor in detecting 24-hour intraocular pressure-related patterns, J. Ophthalmol. (2016) 2016.
- [43] Gbanaibolou Jombo, Yu Zhang, Acoustic-based machine condition monitoring methods and challenges, Eng 4 (1) (2023) 47–79.
- [44] Young Hoon Jung, Seong Kwang Hong, Hee Seung Wang, Jae Hyun Han, Trung Xuan Pham, Hyunsin Park, Junyeong Kim, Sunghun Kang, Chang D. Yoo, Keon Jae Lee, Flexible piezoelectric acoustic sensors and machine learning for speech processing, Adv. Mater. 32 (35) (2020) 1904020.
- [45] Yuma Koizumi, Shoichiro Saito, Hisashi Uematsu, Noboru Harada, Keisuke Imoto, Toyadmos: A dataset of miniature-machine operating sounds for anomalous sound detection, in: 2019 IEEE Workshop on Applications of Signal Processing to Audio and Acoustics, WASPAA, IEEE, 2019, pp. 313–317.
- [46] Kenichi Kumatani, John McDonough, Bhiksha Raj, Microphone array processing for distant speech recognition: From close-talking microphones to far-field sensors, IEEE Signal Process. Mag. 29 (6) (2012) 127–140.
- [47] Kazuki Morita, Tomohiko Yano, Khai Q. Tran, Anomalous sound detection by using local outlier factor and gaussian mixture model, in: Proceedings of the 5th Workshop on Detection and Classification of Acoustic Scenes and Events, DCASE, Tokyo, Japan, 2020, pp. 2–4.
- [48] Dan Hendrycks, Mantas Mazeika, Thomas Dietterich, Deep anomaly detection with outlier exposure, 2018, arXiv preprint arXiv:1812.04606.
- [49] Ali Bou Nassif, Ismail Shahin, Imtinan Attili, Mohammad Azzeh, Khaled Shaalan, Speech recognition using deep neural networks: A systematic review, IEEE Access 7 (2019) 19143–19165.
- [50] Huijun Ding, Xiaolan Shu, Yukun Jin, Taojian Fan, Han Zhang, Recent advances in nanomaterial-enabled acoustic devices for audible sound generation and detection, Nanoscale 11 (13) (2019) 5839–5860.

- [51] Jae Hyun Han, Kang Min Bae, Seong Kwang Hong, Hyunsin Park, Jun-Hyuk Kwak, Hee Seung Wang, Daniel Juhyung Joe, Jung Hwan Park, Young Hoon Jung, Shin Hur, et al., Machine learning-based self-powered acoustic sensor for speaker recognition, Nano Energy 53 (2018) 658–665.
- [52] Zhong Lin Wang, Tao Jiang, Liang Xu, Toward the blue energy dream by triboelectric nanogenerator networks, Nano Energy 39 (2017) 9–23.
- [53] Jae Hyun Han, Jun-Hyuk Kwak, Daniel Juhyung Joe, Seong Kwang Hong, Hee Seung Wang, Jung Hwan Park, Shin Hur, Keon Jae Lee, Basilar membraneinspired self-powered acoustic sensor enabled by highly sensitive multi tunable frequency band, Nano Energy 53 (2018) 198–205.
- [54] Jiangtao Xue, Yang Zou, Yulin Deng, Zhou Li, Bioinspired sensor system for health care and human-machine interaction, EcoMat 4 (5) (2022) e12209.
- [55] N. Suresh Kumar, R. Padma Suvarna, K. Chandra Babu Naidu, Prasun Banerjee, A Ratnamala, H. Manjunatha, A review on biological and biomimetic materials and their applications, Appl. Phys. A 126 (6) (2020) 445.
- [56] Qianming Li, Hu Liu, Shuaidi Zhang, Dianbo Zhang, Xianhu Liu, Yuxin He, Liwei Mi, Jiaoxia Zhang, Chuntai Liu, Changyu Shen, et al., Superhydrophobic electrically conductive paper for ultrasensitive strain sensor with excellent anticorrosion and self-cleaning property, ACS Appl. Mater. Interfaces 11 (24) (2019) 21904–21914.
- [57] Wookey Lee, Jessica Jiwon Seong, Busra Ozlu, Bong Sup Shim, Azizbek Marakhimov, Suan Lee, Biosignal sensors and deep learning-based speech recognition: A review, Sensors 21 (4) (2021) 1399.
- [58] Abdullah Mohammed, Lihui Wang, Brainwaves driven human-robot collaborative assembly, CIRP Ann. 67 (1) (2018) 13–16.
- [59] Anamul Islam, Kenneth Sundaraj, Badlishah Ahmad, Nizam Uddin Ahamed, Asraf Ali, Mechanomyography sensors for muscle assessment: a brief review, J. Phys. Ther. Sci. 24 (12) (2012) 1359–1365.
- [60] Liang Zou, Chang Ge, Z. Jane Wang, Edmond Cretu, Xiaoou Li, Novel tactile sensor technology and smart tactile sensing systems: A review, Sensors 17 (11) (2017) 2653.
- [61] Cheng Chi, Xuguang Sun, Ning Xue, Tong Li, Chang Liu, Recent progress in technologies for tactile sensors, Sensors 18 (4) (2018) 948.
- [62] Pinyo Puangmali, Kaspar Althoefer, Lakmal D. Seneviratne, Declan Murphy, Prokar Dasgupta, State-of-the-art in force and tactile sensing for minimally invasive surgery, IEEE Sens. J. 8 (4) (2008) 371–381.
- [63] Akihiko Yamaguchi, Christopher G. Atkeson, Recent progress in tactile sensing ing and sensors for robotic manipulation: can we turn tactile sensing into vision? Adv. Robot. 33 (14) (2019) 661–673.
- [64] Irvin Hussein Lopez-Nava, Angelica Munoz-Melendez, Wearable inertial sensors for human motion analysis: A review, IEEE Sens. J. 16 (22) (2016) 7821–7834.
- [65] Sara García-de Villa, David Casillas-Pérez, Ana Jiménez-Martín, Juan Jesuś García-Domínguez, Inertial sensors for human motion analysis: A comprehensive review, IEEE Trans. Instrum. Meas. (2023).
- [66] Marion Mundt, William R. Johnson, Wolfgang Potthast, Bernd Markert, Ajmal Mian, Jacqueline Alderson, A comparison of three neural network approaches for estimating joint angles and moments from inertial measurement units, Sensors 21 (13) (2021) 4535.
- [67] Tahera Kalsoom, Naeem Ramzan, Shehzad Ahmed, Masood UrRehman, Advances in sensor technologies in the era of smart factory and industry 4.0, Sensors 20 (23) (2020) 6783.
- [68] K Maharana, S Mondal, B. Nemade, A review: Data pre-processing and data augmentation techniques, Glob. Transit. Proc. 3 (1) (2022) 91–99.
- [69] J.F Davis, M.J Piovoso, K.A Hoo, B.R. Bakshi, Process data analysis and interpretation, in: Advances in Chemical Engineering, Vol. 25, Academic Press, 1999, pp. 1–103.
- [70] N Shi, S Guo, R. Al Kontar, Personalised feature extraction for manufacturing process signature characterisation and anomaly detection, J. Manuf. Syst. 74 (2024) 435–448.
- [71] P Karuppusamy, I Perikos, F Shi, T.N. Nguyen, Sustainable communication networks and application, Lect. Notes Data Eng. Commun. Technol. (2020) 65–72.
- [72] A.O Salau, S. Jain, Feature extraction: a survey of the types, techniques, applications, in: 2019 International Conference on Signal Processing and Communication, ICSC, IEEE, 2019, pp. 158–164.
- [73] H Durrant-Whyte, T.C. Henderson, Multisensor data fusion, in: Springer Handbook of Robotics, 2016, pp. 867–896.
- [74] Mohammed Laroui, Boubakr Nour, Hassine Moungla, Moussa A. Cherif, Hossam Afifi, Mohsen Guizani, Edge and fog computing for iot: A survey on current research activities & future directions, Comput. Commun. 180 (2021) 210–231.
- [75] Jayavardhana Gubbi, Rajkumar Buyya, Slaven Marusic, Marimuthu Palaniswami, Internet of things (iot): A vision, architectural elements, and future directions, Future Gener. Comput. Syst. 29 (7) (2013) 1645–1660, Including Special sections: Cyber-enabled Distributed Computing for Ubiquitous Cloud and Network Services & Cloud Computing and Scientific Applications Big Data, Scalable Analytics, and Beyond.
- [76] Pintu Kumar Sadhu, Venkata P. Yanambaka, Ahmed Abdelgawad, Internet of things: Security and solutions survey, Sensors 22 (19) (2022) 7433.
- [77] Aditya Gaur, Bryan Scotney, Gerard Parr, Sally McClean, Smart city architecture and its applications based on IoT, Procedia Comput. Sci. 52 (2015) 1089–1094.

- [78] Rong Du, Paolo Santi, Ming Xiao, Athanasios V. Vasilakos, Carlo Fischione, The sensable city: A survey on the deployment and management for smart city monitoring, IEEE Commun. Surv. Tutor. 21 (2) (2018) 1533–1560.
- [79] Pintu Kumar Sadhu, Venkata P. Yanambaka, Ahmed Abdelgawad, Kumar Yelamarthi, Prospect of internet of medical things: A review on security requirements and solutions, Sensors 22 (15) (2022) 5517.
- [80] Hang Zhang, Bo Liu, Hongyu Wu, Smart grid cyber–physical attack and defense: A review, IEEE Access 9 (2021) 29641–29659.
- [81] Paulo Shakarian, Stuxnet: Cyberwar revolution in military affairs, 2011.
- [82] Gaoqi Liang, Steven R. Weller, Junhua Zhao, Fengji Luo, Zhao Yang Dong, The 2015Ukraine blackout: Implications for false data injection attacks, IEEE Trans. Power Syst. 32 (4) (2016) 3317–3318.
- [83] Pintu Kumar Sadhu, Venkata P. Yanambaka, Saraju P. Mohanty, Elias Kougianos, Easy-sec: Puf-based rapid and robust authentication framework for the Internet of vehicles, 2022, arXiv preprint arXiv:2204.07709.
- [84] Palak Bagga, Ashok Kumar Das, Mohammad Wazid, Joel JPC. Rodrigues, Youngho Park, Authentication protocols in the internet of vehicles: Taxonomy, analysis, and challenges, Ieee Access 8 (2020) 54314–54344.
- [85] Emiliano Sisinni, Abusayeed Saifullah, Song Han, Ulf Jennehag, Mikael Gidlund, Industrial internet of things: Challenges, opportunities, and directions, IEEE Trans. Ind. Inform. 14 (11) (2018) 4724–4734.
- [86] Mahdi H. Miraz, Maaruf Ali, Peter S. Excell, Rich Picking, Picking A review on internet of things (IoT), Internet of everything (IOE) and internet of nano things (iont), in: 2015 Internet Technologies and Applications, ITA, 2015, pp. 219–224.
- [87] Mikael Gidlund, Tomas Lennvall, Johan Åkerberg, Will 5 g become yet another wireless technology for industrial automation? in: 2017 IEEE International Conference on Industrial Technology, ICIT, IEEE, 2017, pp. 1319–1324.
- [88] Bluetooth Core Specification, Bluetooth Core Specification 5.0, Kirkland, WA, USA, 2016.
- [89] Gaetano Patti, Luca Leonardi, Lucia Lo Bello, A bluetooth low energy realtime protocol for industrial wireless mesh networks, in: IECON 201642nd Annual Conference of the IEEE Industrial Electronics Society, IEEE, 2016, pp. 4627–4632.
- [90] Mauro Marinoni, Alessandro Biondi, Pasquale Buonocunto, Gianluca Franchino, Daniel Cesarini, Giorgio Buttazzo, Real-time analysis and design of a dual protocol support for bluetooth le devices, IEEE Trans. Ind. Inform. 13 (1) (2016) 80–91.
- [91] Baiqing Zong, Chen Fan, Xiyu Wang, Xiangyang Duan, Baojie Wang, Jianwei Wang, 6 g technologies: Key drivers, core requirements, system architectures, and enabling technologies, IEEE Veh. Technol. Mag. 14 (3) (2019) 18–27.
- [92] Ian F. Akyildiz, Ahan Kak, Shuai Nie, 6 g and beyond: The future of wireless communications systems, IEEE Access 8 (2020) 133995–134030.
- [93] Viviane Cunha Farias da Costa, Luiz Oliveira, Jano de Souza, Internet of everything (ioe) taxonomies: A survey and a novel knowledge-based taxonomy, Sensors 21 (2) (2021) 568.
- [94] Peter Mell, Tim Grance, et al., The nist definition of cloud computing, 2011.
- [95] Prince Kwame Senyo, John Effah, Erasmus Addae, Preliminary insight into cloud computing adoption in a developing country, J. Enterpr. Inf. Manage. (2016).
- [96] Bader Alouffi, Muhammad Hasnain, Abdullah Alharbi, Wael Alosaimi, Hashem Alyami, Muhammad Ayaz, A systematic literature review on cloud computing security: threats and mitigation strategies, IEEE Access 9 (2021) 57792–57807.
- [97] P Velmurugadass, S Dhanasekaran, S. Shasi Anand, V. Vasudevan, Enhancing blockchain security in cloud computing with IoT environment using ecies and cryptography hash algorithm, Mater. Today: Proc. 37 (2021) 2653–2659.
- [98] Andrzej Wilczynśki, Joanna Kol odziej, Modelling and simulation of securityaware task scheduling in cloud computing based on blockchain technology, Simul. Model. Pract. Theory 99 (2020) 102038.
- [99] Mohammad Aazam, Eui-Nam Huh, Fog computing: The cloudiot\/ioe middleware paradigm, IEEE Potentials 35 (3) (2016) 40–44.
- [100] Shanhe Yi, Zijiang Hao, Zhengrui Qin, Qun Li, Fog computing: Platform and applications, in: 2015 Third IEEE Workshop on Hot Topics in Web Systems and Technologies (HotWeb), IEEE, 2015, pp. 73–78.
- [101] Flavio Bonomi, Rodolfo Milito, Jiang Zhu, Sateesh Addepalli, Fog computing and its role in the Internet of Things, in: Proceedings of the First Edition of the MCC Workshop on Mobile Cloud Computing, 2012, pp. 13–16.
- [102] Kai Fan, Huiyue Xu, Longxiang Gao, Hui Li, Yintang Yang, Efficient and privacypreserving access control scheme for fog-enabled IoT, Future Gener. Comput. Syst. 99 (2019) 134–142.
- [103] Jialu Yang, Zheyuan Chen, Ying Liu, Michael Ryan, Sliding window filter-based strip breakage modelling for failure prediction, in: 2021 IEEE 17th International Conference on Automation Science and Engineering, CASE, IEEE, 2021, pp. 1461–1466.
- [104] Thanasis Kotsiopoulos, Panagiotis Sarigiannidis, Dimosthenis Ioannidis, Dimitrios Tzovaras, Machine learning and deep learning in smart manufacturing: The smart grid paradigm, Comp. Sci. Rev. 40 (2021) 100341.
- [105] Chao Ning, Fengqi You, Optimization under uncertainty in the era of big data and deep learning: When machine learning meets mathematical programming, Comput. Chem. Eng. 125 (2019) 434–448.

- [106] A Sharma, Z Zhang, R. Rai, The interpretive model of manufacturing: a theoretical framework and research agenda for machine learning in manufacturing, Int. J. Prod. Res. 59 (16) (2021) 4960–4994.
- [107] C Gobert, E.W Reutzel, J Petrich, A.R Nassar, S. Phoha, Application of supervised machine learning for defect detection during metallic powder bed fusion additive manufacturing using high-resolution imaging, Addit. Manuf. 21 (2018) 517–528.
- [108] D Weichert, P Link, A Stoll, S Rüping, S Ihlenfeldt, S. Wrobel, A review of machine learning for the optimization of production processes, Int. J. Adv. Manuf. Technol. 104 (5) (2019) 1889–1902.
- [109] D.H Kim, T.J Kim, X Wang, M Kim, Y.J Quan, J.W Oh, S.H Min, H Kim, B Bhandari, I Yang, S.H. Ahn, Smart machining process using machine learning: A review and perspective on machining industry, Int. J. Precis. Eng. Manuf.-Green Technol. 5 (2018) 555–568.
- [110] J.V Abellan-Nebot, F. Romero Subirón, A review of machining monitoring systems based on artificial intelligence process models, Int. J. Adv. Manuf. Technol. 47 (2010) 237–257.
- [111] M Chandrasekaran, M Muralidhar, C.M Krishna, U.S. Dixit, Application of soft computing techniques in machining performance prediction and optimization: a literature review, Int. J. Adv. Manuf. Technol. 46 (2010) 445–464.
- [112] S Dutta, S.K Pal, S Mukhopadhyay, R. Sen, Application of digital image processing in tool condition monitoring: A review, CIRP J. Manuf. Sci. Technol. 6 (3) (2013) 212–232.
- [113] Y.H Kuo, A. Kusiak, From data to big data in production research: the past and future trends, Int. J. Prod. Res. 57 (15–16) (2019) 4828–4853.
- [114] E Oztemel, S. Gursev, Literature review of Industry 4.0 and related technologies, J. Intell. Manuf. 31 (1) (2020) 127–182.
- [115] T Kotsiopoulos, P Sarigiannidis, D Ioannidis, D. Tzovaras, Machine learning and deep learning in smart manufacturing: The smart grid paradigm, Comp. Sci. Rev. 40 (2021) 100341.
- [116] P Henderson, R Islam, P Bachman, J Pineau, D Precup, D. Meger, Deep reinforcement learning that matters, in: Proceedings of the AAAI Conference on Artificial Intelligence, Vol. 32, 2018, No. 1.
- [117] C Li, P Zheng, Y Yin, B Wang, L. Wang, Deep reinforcement learning in smart manufacturing: A review and prospects, CIRP J. Manuf. Sci. Technol. 40 (2023) 75–101.
- [118] J Wang, Y Ma, L Zhang, R.X Gao, D. Wu, Deep learning for smart manufacturing: Methods and applications, J. Manuf. Syst. 48 (2018) 144–156.
- [119] Celestine Iwendi, Zunera Jalil, Abdul Rehman Javed, Thippa Reddy, Rajesh Kaluri, Gautam Srivastava, Ohyun Jo, Keysplitwatermark: Zero watermarking algorithm for software protection against cyber-attacks, IEEE Access 8 (2020) 72650–72660.
- [120] Mamoona Majid, Shaista Habib, Abdul Rehman Javed, Muhammad Rizwan, Gautam Srivastava, Thippa Reddy Gadekallu, Jerry ChunWei Lin, Applications of wireless sensor networks and internet of things frameworks in the industry revolution 4.0: A systematic literature review, Sensors 22 (6) (2022) 2087.
- [121] Mojtaba Jamshidi, Ehsan Zangeneh, Mehdi Esnaashari, Aso Mohammad Darwesh, Mohammad Reza Meybodi, A novel model of Sybil attack in cluster-based wireless sensor networks and propose a distributed algorithm to defend it, Wirel. Pers. Commun. 105 (2019) 145–173.
- [122] Dong-Seong Kim, Hoa Tran-Dang, Dong-Seong Kim, Hoa Tran-Dang, An overview on industrial control networks, Ind. Sens. Controls Commun. Netw.: Wired Technol. Cloud Comput. Internet Things (2019) 3–16.
- [123] Dionisis Kandris, Christos Nakas, Dimitrios Vomvas, Grigorios Koulouras, Applications of wireless sensor networks: an up-to-date survey, Appl. Syst. Innov. 3 (1) (2020) 14.
- [124] Ian F Akyildiz, Weilian Su, Yogesh Sankarasubramaniam, Erdal Cayirci, Wireless sensor networks: a survey, Comput. Netw. 38 (4) (2002) 393–422.
- [125] Th Arampatzis, John Lygeros, Stamatis Manesis, A survey of applications of wireless sensors and wireless sensor networks, in: Proceedings of the 2005 IEEE International Symposium on, Mediterrean Conference on Control and Automation Intelligent Control, 2005, IEEE, 2005, pp. 719–724.
- [126] Behailu Negash, Amir M. Rahmani, Pasi Liljeberg, Axel Jantsch, Fog computing fundamentals in the internet-of-things, in: Fog Computing in the Internet of Things: Intelligence at the Edge, 2018, pp. 3–13.
- [127] Arsalan Khan, Sumair Aziz, Mudassar Bashir, Muhammad Umar Khan, IoT and wireless sensor network-based autonomous farming robot, in: 2020 International Conference on Emerging Trends in Smart Technologies, ICETST, IEEE, 2020, pp. 1–5.
- [128] Chi-Tung Chen, Cheng-Chi Lee, Iuon-Chang Lin, Efficient and secure three-party mutual authentication key agreement protocol for WSNs in IoT environments, PLoS One 15 (4) (2020) e0232277.
- [129] Antonio Forcina, Domenico Falcone, The role of industry 4.0 enabling technologies for safety management: A systematic literature review, Procedia Comput. Sci. 180 (2021) 436–445.
- [130] Beata Mrugalska, Junaid Ahmed, Organizational agility in industry 4.0: A systematic literature review, Sustainability 13 (15) (2021) 8272.
- [131] Sehrish Atif, Shehzad Ahmed, Muhammad Wasim, Bassam Zeb, Zeeshan Pervez, Lorraine Quinn, Towards a conceptual development of industry 4.0, servitisation, and circular economy: A systematic literature review, Sustainability 13 (11) (2021) 6501.

- [132] Wei Jiang, Bin Han, Mohammad Asif Habibi, Hans Dieter Schotten, The road towards 6 g: A comprehensive survey, IEEE Open J. Commun. Soc. 2 (2021) 334–366.
- [133] Cheng-Xiang Wang, Xiaohu You, Xiqi Gao, Xiuming Zhu, Zixin Li, Chuan Zhang, Haiming Wang, Yongming Huang, Yunfei Chen, Harald Haas, et al., On the road to 6 g: Visions, requirements, key technologies and testbeds, IEEE Commun. Surv. Tutor. (2023).
- [134] Kuïşat Tekbıyık, Ali Rıza Ekti, Guñeş Karabulut Kurt, Ali Görçin, Terahertz band communication systems: Challenges, novelties and standardization efforts, Phys. Commun. 35 (2019) 100700.
- [135] Mostafa Zaman Chowdhury, Moh Khalid Hasan, Md. Shahjalal, Md. Tanvir Hossan, Yeong Min Jang, Optical wireless hybrid networks: Trends, opportunities, challenges, and research directions, IEEE Commun. Surv. Tutor. 22 (2) (2020) 930–966.
- [136] Georgia Kolovou, Sharief Oteafy, Periklis Chatzimisios, A remote surgery use case for the ieee p1918. 1 tactile internet standard, in: ICC 2021-IEEE International Conference on Communications, IEEE, 2021, pp. 1–6.
- [137] Wei Zhang, Dong-Sheng Ding, Yu-Bo Sheng, Lan Zhou, Bao-Sen Shi, Guang-Can Guo, Quantum secure direct communication with quantum memory, Phys. Rev. Lett. 118 (22) (2017) 220501.
- [138] Ali Dorri, Marco Steger, Salil S. Kanhere, Raja Jurdak, Blockchain: A distributed solution to automotive security and privacy, IEEE Commun. Mag. 55 (12) (2017) 119–125.
- [139] Pawani Porambage, Gurkan Gur, Diana Pamela Moya Osorio, Madhusanka Liyanage, Andrei Gurtov, Mika Ylianttila, The roadmap to 6 g security and privacy, IEEE Open J. Commun. Soc. 2 (2021) 1094–1122.
- [140] Oliver Holland, Eckehard Steinbach, R. Venkatesha Prasad, Qian Liu, Zaher Dawy, Adnan Aijaz, Nikolaos Pappas, Kishor Chandra, Vijay S. Rao, Sharief Oteafy, et al., The ieee 1918.1 tactile internet standards working group and its standards, Proc. IEEE 107 (2) (2019) 256–279.
- [141] Walid Saad, Mehdi Bennis, Mingzhe Chen, A vision of 6 g wireless systems: Applications, trends, technologies, and open research problems, IEEE Netw. 34 (3) (2019) 134–142.
- [142] Paul M. Fitts, Human engineering for an effective air-navigation and traffic-control system, 1951.
- [143] Joost CF. De Winter, Riender Happee, Marieke H Martens, Neville A. Stanton, Effects of adaptive cruise control and highly automated driving on workload and situation awareness: A review of the empirical evidence, Transp. Res. F 27 (2014) 196–217.
- [144] Christian P Janssen, Linda Ng Boyle, Andrew L Kun, Wendy Ju, Lewis L. Chuang, A hidden Markov framework to capture human-machine interaction in automated vehicles, Int. J. Hum.-Comput. Interact. 35 (11) (2019) 947–955.
- [145] Kim J Vicente, Emilie M Roth, Randall J. Mumaw, How do operators monitor a complex, dynamic work domain? The impact of control room technology, Int. J. Hum.-Comput. Stud. 54 (6) (2001) 831–856.
- [146] Shadan Sadeghian Borojeni, Lars Weber, Wilko Heuten, Susanne Boll, From reading to driving: priming mobile users for take-over situations in highly automated driving, in: Proceedings of the 20th International Conference on Human–Computer Interaction with Mobile Devices and Services, 2018, pp. 1–12.
- [147] Christopher DD. Cabrall, Alexander Eriksson, Felix Dreger, Riender Happee, Joost de Winter, How to keep drivers engaged while supervising driving automation? A literature survey and categorisation of six solution areas, Theor. Issues Ergon. Sci. 20 (3) (2019) 332–365.
- [148] Natasha Merat, A. Hamish Jamson, Frank CH. Lai, Michael Daly, Oliver MJ. Carsten, Transition to manual: Driver behaviour when resuming control from a highly automated vehicle, Transp. Res. F 27 (2014) 274–282.
- [149] Donald A. Norman, The design of future things: Cautious cars, in: Driving Assessment Conference, Vol. 4, University of Iowa, 2007.
- [150] John D Lee, Katrina A., See. Trust in automation: Designing for appropriate reliance, Hum. Factors 46 (1) (2004) 50–80.
- [151] Chao Zhang, Zenghui Wang, Guanghui Zhou, Fengtian Chang, Dongxu Ma, Yanzhen Jing, Wei Cheng, Kai Ding, Dan Zhao, Towards new generation humancentric smart manufacturing in industry 5.0: A systematic review, Adv. Eng. Inform. 57 (2023) 102121.
- [152] Paul Kengfai Wan, Torbjørn Langedahl Leirmo, Human-centric zero defect manufacturing: State-of-the-art review, perspectives, and challenges, Comput. Ind. 144 (2023) 103792.
- [153] Patrick Rannertshauser, Melanie Kessler, Julia C. Arlinghaus, Human-centricity in the design of production planning and control systems: A first approach towards industry 5.0, IFAC-PapersOnLine 55 (10) (2022) 2641–2646.
- [154] Iveta Zolotová, Peter Papcun, Erik Kajáti, Martin Mišskuf, Jozef Mocnej, Smart and cognitive solutions for operator 4.0: Laboratory h-cpps case studies, Comput. Ind. Eng. 139 (2020) 105471.
- [155] Sandra Grabowska, Smart factories in the age of Industry 4.0, Manage. Syst. Prod. Eng. 2 (28) (2020) 90–96.
- [156] P.L Morgan, A Voinescu, C Williams, P Caleb-Solly, C Alford, I Shergold, G Parkhurst, A. Pipe, An emerging framework to inform effective design of human-machine interfaces for older adults using connected autonomous vehicles, in: Advances in Human Aspects of Transportation: Proceedings of

the AHFE 2017 International Conference on Human Factors in Transportation, July (2017) 17-21, Vol. 8, Springer International Publishing, The Westin Bonaventure Hotel, Los Angeles, California, USA, 2018, pp. 325–334.

- [157] B Shneiderman, C. Plaisant, Designing the User Interface: Strategies for Effective Human–Computer Interaction, Pearson Education India, 2010.
- [158] C.D Wickens, S.E Gordon, Y Liu, J. Lee, An Introduction to Human Factors Engineering, Vol. 2, Pearson Prentice Hall, Upper Saddle River, NJ, 2004, p. 587.
- [159] H.M. Cuevas, An illustrative example of four HCI design approaches for evaluating an automated system interface, in: Proceedings of the Human Factors and Ergonomics Society Annual Meeting, Vol. 48, SAGE Publications, Sage CA: Los Angeles, CA, 2004, pp. 892–896, No. 5.
- [160] F Tao, H Zhang, A Liu, A.Y. Nee, Digital twin in industry: State-of-the-art, IEEE Trans. Ind. Inform. 15 (4) (2018) 2405–2415.
- [161] M Grieves, J. Vickers, Digital twin: Mitigating unpredictable, undesirable emergent behaviour in complex systems, Transdiscip. Perspect. Complex Syst.: New Find. Appr. (2017) 85–113.
- [162] T.A.O Fei, S.U.N Xuemin, J Cheng, Z.H.U Yonghuai, L.I.U Weiran, W.A.N.G Yong, X.U Hui, H.U Tianliang, L.I.U Xiaojun, L.I.U Tingyu, S.U.N. Zheng, MakeTwin: A reference architecture for digital twin software platform, Chin. J. Aeronaut. 37 (1) (2024) 1–18.
- [163] X Yao, N Ma, J Zhang, K Wang, E Yang, M. Faccio, Enhancing wisdom manufacturing as industrial metaverse for industry and society 5.0, J. Intell. Manuf. 35 (1) (2024) 235–255.
- [164] A Siyaev, G.S. Jo, Neuro-symbolic speech understanding in aircraft maintenance metaverse, Ieee Access 9 (2021) 154484–154499.
- [165] W Xian, K Yu, F Han, L Fang, D He, Q.L. Han, Advanced manufacturing in industry 5.0: A survey of key enabling technologies and future trends, IEEE Trans. Ind. Inform. (2023).
- [166] P. De Giovanni, Sustainability of the Metaverse: A transition to Industry 5.0, Sustainability 15 (7) (2023) 6079.
- [167] Z Zheng, T Li, B Li, X Chai, W Song, N Chen, Y Zhou, Y Lin, R. Li, Industrial metaverse: connotation, features, technologies, applications and challenges, in: Asian Simulation Conference, Springer Nature Singapore, Singapore, 2022, pp. 239–263.
- [168] Y Torres, S Nadeau, K. Landau, Evaluation of fatigue and workload among workers conducting complex manual assembly in manufacturing, IISE Trans. Occup. Ergon. Hum. Factors 9 (1) (2021) 49–63.
- [169] S Xu, N.G. Hall, Fatigue, personnel scheduling and operations: Review and research opportunities, European J. Oper. Res. 295 (3) (2021) 807–822.
- [170] A Baghdadi, L.A Cavuoto, A Jones-Farmer, S.E Rigdon, E.T Esfahani, F.M. Megahed, Monitoring worker fatigue using wearable devices: A case study to detect changes in gait parameters, J. Qual. Technol. 53 (1) (2021) 47–71.
- [171] J.F Arinez, Q Chang, R.X Gao, C Xu, J. Zhang, Artificial intelligence in advanced manufacturing: Current status and future outlook, J. Manuf. Sci. Eng. 142 (11) (2020) 110804.
- [172] A Reiman, J Kaivo-oja, E Parviainen, E.P Takala, T. Lauraeus, Human factors and ergonomics in manufacturing in the industry 4.0 context–A scoping review, Technol. Soc. 65 (2021) 101572.
- [173] A Papetti, M Ciccarelli, A Brunzini, M. Germani, Design of ergonomic manufacturing equipment by a human-centered methodology, Int. J. Interact. Des. Manuf. (IJIDEM) 15 (1) (2021) 107–111.
- [174] M.F Mridha, S.C Das, M.M Kabir, A.A Lima, M.R Islam, Y. Watanobe, Braincomputer interface: Advancement and challenges, Sensors 21 (17) (2021) 5746.
- [175] S Saha, K.A Mamun, K Ahmed, R Mostafa, G.R Naik, S Darvishi, A.H Khandoker, M. Baumert, Progress in brain-computer interface: Challenges and opportunities, Front. Syst. Neurosci. 15 (2021) 578875.
- [176] R Bousseta, I. El Ouakouak, M Gharbi, F. Regragui, EEG-based brain-computer interface for controlling a robot arm movement through thought, Irbm 39 (2) (2018) 129–135.
- [177] M Shim, H.J Hwang, D.W Kim, S.H Lee, C.H. Im, Machine-learning-based diagnosis of schizophrenia using combined sensor-level and source-level EEG features, Schizophr. Res. 176 (2–3) (2016) 314–319.
- [178] S Jaskó, A Skrop, T Holczinger, T Chován, J. Abonyi, Development of manufacturing execution systems in accordance with industry 4.0 requirements: A review of standard-and ontology-based methodologies and tools, Comput. Ind. 123 (2020) 103300.
- [179] MESA, Manufacturing Enterprise Solutions Association, MESA International, 2021.
- [180] A Shojaeinasab, T Charter, M Jalayer, M Khadivi, O Ogunfowora, N Raiyani, M Yaghoubi, H. Najjaran, Intelligent manufacturing execution systems: A systematic review, J. Manuf. Syst. 62 (2022) 503–522.
- [181] R Liu, B Yang, E Zio, X. Chen, Artificial intelligence for fault diagnosis of rotating machinery: A review, Mech. Syst. Signal Process. 108 (2018) 33–47.
- [182] F Tao, H Zhang, C. Zhang, Advancements and challenges of digital twins in industry, Nat. Comput. Sci. 4 (3) (2024) 169–177.
- [183] J Egger, T. Masood, Augmented reality in support of intelligent manufacturing-a systematic literature review, Comput. Ind. Eng. 140 (2020) 106195.
- [184] Z Ji, Q Liu, W Xu, B Yao, J Liu, Z. Zhou, A closed-loop brain-computer interface with augmented reality feedback for industrial human-robot collaboration, Int. J. Adv. Manuf. Technol. (2021) 1–16.